Workshop to Identify Alternative Energy Environmental Information Needs

Workshop Summary

October 2007





OCS Report MMS 2007-057

Workshop to Identify Alternative Energy Environmental Information Needs

Workshop Summary

Authors

Jacqueline Michel Research Planning, Inc.

Elizabeth Burkhard Environmental Studies Branch, Minerals Management Service

October 2007

Prepared under MMS Contract 1435-01-06-CT-39821

by

Research Planning, Inc. 1121 Park Street Columbia, South Carolina 29201

Published by

MARS U.S. Department of the Interior Minerals Management Service

Disclaimer

This report was prepared under contract between the Minerals Management Service (MMS) and Research Planning, Inc. (RPI). This report has been technically reviewed by the MMS, and it has been approved for publication. Approval does not signify that the contents necessarily reflect the views and policies of the MMS, nor does mention of trade names or commercial products constitute endorsement or recommendation for use.

Citation

Michel, J. and Burkhard, E. 2007. Workshop to Identify Alternative Energy Environmental Information Needs: Workshop Summary. U.S. Department of the Interior, Minerals Management Service, Herndon, VA, MMS OCS Report 2007-057. 50 pp. + appendices.

Acknowledgements

MMS thanks all the participants at the Workshop to Identify Alternative Energy Environmental Information Needs, held on 26-28 June 2007 in Herndon, VA. We were pleased that there was such interest and discussion. A special note of thanks is extended to each of the presenters.

The success of the breakout sessions was due to the efforts of the MMS leads Elizabeth Burkhard, Rodney Cluck, Maurice Hill, and Jeff Ji as well as the facilitators Sean M. Casey, Jacqueline Michel, John Poulin, and Scott Graves. The dedication of the note takers, Alexis Castrovinci, Erin Healy, Leslie Safier, and Samuel Edelstein, is gratefully acknowledged. Wendy Early of Research Planning, Inc. did an excellent job as the workshop coordinator.

Table of Contents

List of	Abbrev	viations and Acronyms	iii
1.0	Introdu	action	1
	1.1	Purpose and Intent of the Workshop	1
	1.2	Workshop Format	2
2.0	Techni	cal Presentations	3
3.0	Aquati	c Resources Breakout Group	7
4.0	Flying	Animals Breakout Group	19
5.0	Physic	al Oceanography and Air Quality Breakout Group	29
6.0	Social	Sciences and Economics	37
7.0	Final F	Plenary Session	48
	7.1	Geospatial Data and Tools	48
	7.2	Geospatial Analytical Tools	48
	7.3	Evaluation of Mitigation Measure	49
	7.4	Development of Guidance Documents and Study Protocols for Impact	
		Assessments	49
	7.5	Overarching Messages and Conclusion	49
	1		

Appendix A:	Workshop Agenda
Appendix B:	List of Participants

- Appendix C:Presentation Slides for Each SpeakerAppendix D:Biographies for Invited Speakers

List of Tables

1	Summary of comments on information needs for EMF – attraction and avoidance	
	effects on aquatic resources1	0
2	Summary of comments on information needs for noise – human and animal	
	effects on aquatic resources1	1
3	Summary of comments on information needs for migratory issues (local, long distance,	
	vertical) on aquatic resources1	2
4	Summary of comments on information needs for aquatic threatened and endangered	
	species1	3
5	Summary of comments on information needs for collision, entanglement, and	
	entrainment for aquatic resources1	4
6	Summary of comments on information needs for habitat changes for	
	aquatic resources1	5
7	Summary of comments on information needs for attraction and avoidance impacts for	
	aquatic resources1	6

8	Summary of comments on information needs for impacts of contaminants on aquatic resources	.16
9	Summary of comments on information needs for regional summaries of bird/bat abundance data for priority offshore areas	.22
10	Summary of comments on information needs for regional baseline studies in priority areas to fill data gaps for birds/bats	.23
11	Summary of comments on information needs for synthesis of old and new flying anima data into GIS	ıl .23
12	Summary of comments on information needs for site-specific study protocols and guidelines: Collision/mortality detection	.24
13	Summary of comments on information needs for site-specific study protocols and guidelines: Attraction/Avoidance	.25
14	Summary of comments on information needs for improved predictive models	.25
15	Summary of comments on information needs for mitigation measures evaluation	.26
16	Summary of comments on information needs for a risk assessment framework to address key issues	.27
17	Summary of comments on information needs for geospatial data tools to support energy resource mapping and physical processes impact assessments	.31
18	Summary of comments on information needs for developing a baseline of existing conditions in the nearfield.	.32
19	Summary of comments on information needs for developing a baseline of existing conditions in the farfield	.32
20	Summary of comments on information needs for assessing the nearfield impacts of structures	.33
21	Summary of comments on information needs for assessing the farfield impacts of structures	.34
22	Summary of comments on information needs for how to apply quantitative impacts data on physical processes	.34
23	Summary of comments on information needs for how to apply adaptive management principles to monitoring and mitigation requirements for alternative energy installations	.40
24	Summary of comments on information needs for treatment of commercial versus experimental technologies	.41
25	Summary of comments on information needs for developing cost/benefit methodologie for alternative energy projects	s .42
26	Summary of comments on information needs for space-use conflicts and mapping	.44
27	Summary of comments on information needs for understanding stakeholders	.45
28	Summary of comments on information needs for regional and national issues	.46
29	Summary of comments on information needs for regulatory collaboration	.46

List of Abbreviations and Acronyms

ADCP	acoustic doppler current profiler
BACI	before after control impact
CMACS	Centre for Marine and Coastal Studies, Ltd.
CODAR	Coastal Ocean Dynamics Applications Radar
COWRIE	Collaborative Offshore Wind Research into the Environment
EIA	Environmental Impact Assessment
EIS	Environmental Impact Study
EMF	electromagnetic fields
EPRI	Electric Power Research Institute
ESP	Environmental Studies Program
FERC	Federal Energy Regulatory Commission
FLOWW	Fisheries Liaison for Offshore Wind and Wet
GIS	geographic information systems
ILP	Integrated Licensing Process
MMS	Minerals Management Service
MW	megawatts
NGO	non-governmental organization
NIMBY	not in my back yard
NMFS	National Marine Fisheries Service
NOAA	National Oceanic and Atmospheric Administration
NODC	National Oceanographic Data Center
NOREL	Nautical Offshore Renewable Energy Liaison Group
NREL	National Renewable Energy Laboratory
NWCC	National Wind Coordinating Committee
OCS	Outer Continental Shelf
OMB	Office of Management and Budget
OREEF	Offshore Renewable Energy and Environment Forum
R&D	research and development
RAG	Research Advisory Group
SEA	Strategic Environmental Assessment
UK	United Kingdom
U.S.	United States
USACE	U.S. Army Corps of Engineers
USEPA	U.S. Environmental Protection Agency
USFWS	U.S. Fish and Wildlife Service
USGS	U.S. Geological Survey

1.0 Introduction

1.1 Purpose and Intent of the Workshop

With the passage of the Energy Policy Act of 2005, the Minerals Management Service (MMS) assumed regulatory responsibilities on the Outer Continental Shelf (OCS) for activities that produce or support the production, transportation, or transmission of energy from sources other than oil and gas (i.e., alternative energy). Potential impacts on the human and marine environments must be evaluated in order for MMS to make environmentally sound decisions when authorizing alternative energy activities on the OCS. MMS funded a synthesis and analysis report that reviewed existing data on environmental effects of alternative energy uses and identified information needs, entitled *Worldwide Synthesis and Analysis of Existing Information Regarding Environmental Effects of Alternative Energy Uses on the Outer Continental Shelf* (Michel et al., 2007). This report provided the basis for a workshop, attended by experts in the field, to identify data needs and to outline potential studies for the MMS Environmental Studies Program and its partners.

The workshop was held on 26-28 June 2007 in Herndon, Virginia. Over 200 invitation letters were sent out to a broad mix of representatives of Federal and State agencies, industry, academia, non-governmental organizations (NGOs), and national and international experts. Participants registered on a workshop website, which was updated frequently with the list of current registrants. In all, 144 people attended the workshop.

This workshop was an important step for the MMS in communicating and developing a collaborative relationship with other Federal agencies, affected State and local groups, and industry. Members of groups with knowledge about existing offshore alternative energy development (i.e., academia or representatives from countries or states where development is already occurring) were invited to share their expertise.

The intent of the workshop was to:

- Gather and initiate a dialog among representatives from within the MMS, other Federal agencies, States where development is most likely to occur, the alternative energy industry, academics, and foreign experts in the field.
- Evaluate the body of information currently available.
- Identify data and knowledge needs in the information available about the human and marine environments with respect to top alternative energy issues.
- Develop a list of potential environmental studies to fill those identified gaps.
- Discuss development of partnerships for information sharing and financial collaboration.

1.2 Workshop Format

The format for the workshop was as follows (see Appendix A for the workshop agenda). The workshop began with ten <u>technical presentations</u> by national and international experts covering the state of wind and wave technologies, future trends, environmental concerns, and lessons learned. The presentations provided a strong technical background for later discussions.

During sign-in, each participant registered for one of four <u>breakout groups</u>: 1) Aquatic Resources; 2) Flying Animals; 3) Physical Oceanography and Air Quality; and 4) Social Sciences and Economics. In each breakout group, a MMS scientist was the technical lead and provided a summary of the information needs for each resource area based on the Synthesis Report. A facilitator guided each group's discussion and asked for a volunteer to report out on behalf of the group in a plenary session at the end of the second day. There were two sessions for each breakout group: first, the groups were asked to discuss and develop a list of critical information needs that were scientifically feasible; after a break, each group was asked to provide supporting detailed information for the most important priorities.

The facilitators briefly reviewed rules and goals of the sessions. During the first session of the breakouts, there was open discussion on the key information needs. The aim was to generate potential project ideas and identify information gaps, not to evaluate them at this point. As the groups reached consensus on what were the priority research needs, the needs were recorded on large-scale posters that were printed with the following subheadings:

- Data Needs
- Suggested Methods
- Collaborators
- Other Limitations
- Special Considerations

Members in each breakout group were given sticky notes to write down specific suggestions and comments under the subheadings for each priority information need. After this exercise, the facilitators led a discussion of each priority information need, based on the posted comments, to further refine and define the topics. At the end of the second breakout session, the MMS scientist, facilitator, and group spokesperson compiled the results of the group into a presentation to be made during the afternoon plenary session. On the morning of the third day, there was a final <u>Plenary Session</u> where the group discussed mutual interest in, and opportunities for, collaboration and partnership on studies.

The results of the workshop are provided in the following sections. Section 2 contains short summaries of each technical presentation. Section 3 contains the results of the Aquatic Resources Breakout Session. Section 4 contains the results of the Flying Animals Breakout Session. Section 5 contains the results of the Physical Oceanography and Air Quality Breakout Session. Section 6 contains the results of the Social Sciences and Economics Breakout Session. Section 7 contains a summary of the discussions during the final Plenary Session. The workshop agenda is in Appendix A. The list of participants who attended the workshop is in Appendix B. The technical presentations are in Appendix C. Biographies for the invited speakers are in Appendix D.

2.0 Technical Presentations

There were eight technical presentations on the first day and two on the morning of the second day (see Agenda in Appendix A). The slides for each powerpoint presentation are included in Appendix C. Following is a summary of each technical presentation.

Current State and Future Trends in Alternative Energy – The Contractors Point of View Kurt Thomsen, *Advanced Offshore Solutions*

The current state of offshore wind technology includes turbines that are 3 megawatts (MW) that are installed in water depths up to 27 meters. The limiting factors are the costs of foundation installation and maintenance in deeper water and further offshore, cabling costs, and transfer power losses with distance offshore. Because of the push to place wind parks further offshore, turbines must become larger to be more cost-effective; the 5 MW turbine will become the standard. Currently, contractors cannot cope with larger turbines and foundations. Thus, there needs to be growth and development in supporting technologies for construction, installation, and maintenance for the larger turbines to be placed further offshore.

Offshore Wind Power in Denmark

Steffen Nielsen, Danish Energy Authority

Denmark currently has 3,100 MW of wind power, providing 20% of the national electricity demand, of which 423 MW is from offshore. Because of plans for up to 4,600 MW from offshore development, the government has set up a "one-stop shop" that coordinates among agencies for consent and approvals. Danish policymakers are also working to reduce financial risks through various programs. One of the biggest challenges is grid integration. The results of monitoring studies at the two large offshore wind parks (Horns Rev and Nysted) were reviewed. These studies have supported refinement of strategic mapping to support selection of 23 potential sites to meet the planned expansion offshore. The studies also indicated the need for better public consultation, to get more local community involvement.

Wind Energy Status and Future

Robert Thresher, National Wind Technology Center, National Renewal Energy Laboratory

The United States currently has 11,600 MW of wind power, all from land-based development. To generate 20% of the electricity demand in the United States, 300 gigawatts would be needed. To support this growth, wind energy costs must be competitive by: increasing turbine size; making significant R&D advances in reliability of components, low wind speed technology, and foundation types; and manufacturing improvements. Further research is also needed to monitor, model, and assess the potential for avian impacts and develop methods and technologies to reduce risks. Wave and ocean current/tidal energy technologies are still in the developmental stages. Modest investments in environmental research and development are needed to reduce potential impacts and thus promote the installation of alternative energy projects offshore.

Offshore Wind in the United Kingdom

Michael Hay, British Wind Energy Association

The United Kingdom (UK) target is 10% electricity from renewable energy sources by 2010 and 20% by 2020. There are plans for 28 sites totaling up to 9,000 MW from offshore wind because of limited space on land and the ability to locate close to demand centers. To accomplish this, the government uses Strategic Environmental Assessments to support scoping studies, developed a simpler consenting process (through the Marine Bill), and is addressing grid issues. There are multiple groups in place to support the development of offshore renewable energy projects including the Collaborative Offshore Wind Research Into the Environment (COWRIE), the Research Advisory Group (RAG), the Offshore Renewable Energy and Environment Forum (OREEF), the Nautical Offshore Renewable Energy Liaison Group (NOREL), and the Fisheries Liaison for Offshore Wind and Wet (FLOWW).

Licensing Offshore Renewable Energy in the UK

Rachel Mills, Marine and Fisheries Agency, Department of the Environment, Food, and Rural Affairs

The license application process in the UK is well established, although there have been many lessons learned. There should be a project management approach to the license application process, with specific targets and a dedicated case worker to provide consistency. Transparency in the process and decisions is essential. Consistency is also needed in license conditions, and monitoring of compliance should be conducted. UK regulators have learned that it is important to make sure that the license conditions clearly specify requirements and are time limited. Otherwise, there can be conflicting views on how the license conditions are interpreted. Another lesson was the need for more flexible and proportionate enforcement; that is, there should be options such as suspension of a license because of violations of the conditions, rather than the only option being revoking of the license.

Impact Assessment and Monitoring of Offshore Wind Farms: UK Perspective

Chris Jenner, RPS Group Plc

The Environmental Impact Assessment (EIA) process for Round 2 in the UK was reviewed as applied to two case studies: The Thames Estuary projects and the Lynn & Inner Dowsing projects. The challenges are to collect baseline data appropriate for impact assessment that focuses on key concerns and specification of conditions to mitigate identified potential impacts. Implementation of consent conditions and validation of the effectiveness of mitigation measures are critical to both existing and future projects.

Wave Energy Development in the United States: Present Status and Future Trends

George Hagerman, Virginia Tech, Advanced Research Institute

Wave energy resources in the United States vary by location and season but are available on all coasts including Alaska and Hawaii. Extracting 15% of total wave energy flux and converting to electricity at 80% efficiency would yield 252 terawatt-hours per year, which is comparable to the annual energy output of all existing conventional hydro-electric projects in the United States.

The technologies for wave energy conversion are diverse, including terminators, attenuators, and point absorbers. There are eleven wave energy projects in the Federal Energy Regulatory Commission (FERC) regulatory process, all on the west coast, and one Navy funded demonstration project in State waters in Hawaii. The learning curve and evolution of designs from initial concept to mature designs that reduce energy costs allows new technologies to be competitive with the market value of the energy.

Environmental Concerns Associated with Wave Energy Conversion Technology in the United States

Greg McMurray, Oregon Department of Land Conservation and Development

Environmental concerns associated with wave energy conversion technologies include wave energy reduction, with estimates of 3-15% reduction, although there are many uncertainties. Shadow effect is minimized if the project is placed more than 3.2 kilometers (2 miles) offshore, although there could be impacts to sediment transport if projects are located within nearshore littoral cells. Hard structures can act as fish attraction devices, potentially creating a collision risk for larger marine mammals and birds. Electromagnetic field risks to listed salmonids are of special concern on the west coast. Chemical toxicity concerns include spills, biofouling coatings, sacrificial anodes, and chemicals in the transmission system. Acoustics impacts may affect individual animals as well as populations if there are ecological barrier effects of the arrays. Space-use conflicts need to be addressed. Cumulative effects are difficult to predict. Major knowledge gaps include array effects, hard substrate effects in a soft-bottom habitat, barrier effects, collision risk for large vertebrates, system survivability, and ecological costs of avoidance. Some of these risks can be evaluated with current information; however, monitoring will be essential to determine actual risks and effectiveness of mitigation measures.

Progress of MMS Regulation Development, Document Production, Scheduling, Government Framework

Maureen Bornholdt, Minerals Management Service

The Energy Policy Act of 2005 amended the OCS Lands Act to permit MMS to authorize alternative energy project activities on the OCS. MMS is establishing an alternative energy program and is in the process of developing comprehensive program regulations. MMS is also processing two individual offshore wind project proposals off the coasts of New York and Massachusetts. The program regulations and development will address interagency coordination, lease issuance, lease administration, project plan reviews, conduct of approved plan activities, and decommissioning. During the lease administration period, the payments, rental details, operating fees, and bonding issues are determined. In addition, a project plan review period (modeled after the UK and Denmark processes) includes a site assessment phase and a construction and operation phase (including geological surveys and operating procedure review). MMS is preparing a Programmatic Environmental Impact Statement, with the final document expected to be released late summer 2007. The Final Rule is expected to be published in 2008.

Highlights of Marine-based Studies to Support Wave, Current, or Tidal Energy Development

Stephen Bowler and Nicholas Jayjack, Federal Energy Regulatory Commission (FERC)

As of 31 May 2007, the FERC Hydropower Program had 38 permits for alternative energy projects in the licensing process, including eight for ocean current technology, four for wave technology, and twenty-six for tidal technology. FERC uses an Integrated Licensing Process (ILP). The Makah Bay wave energy project was reviewed as an example of the implementation of the Alternative Licensing Process and timeline. License conditions include eelgrass and benthic life surveys along the cable route, buoy noise assessment, buoy antifouling paint effectiveness assessment, monitoring of marine mammals to determine entanglement potential, cultural resource impact monitoring during construction, and recreational use monitoring to determine if the buoy array acted as a tourist attraction.

3.0 Aquatic Resources Breakout Group

The Aquatic Resources breakout group covered the following resources: benthic resources, fishery resources, marine mammals, and sea turtles. There were 25 members in the group with broad representation from Federal and State agencies, academia, industry, non-governmental organizations, and the international community. The first session started with the following summary of the information needs identified in the Synthesis Report:

Benthic Resources

- Better benthic habitat mapping of proposed sites for alternative energy facilities.
- Consensus on evaluating the positive or negative effects of the introduction of hard substrates in soft sediments: sustainability, energy flows, species interactions, and an understanding of scale.
- Well-designed monitoring protocols that will provide good baseline data and generate meaningful results at the appropriate level of confidence.
- A definitive study to answer whether electromagnetic fields (EMF) from cables from offshore alternative energy facilities directly affect benthic communities.
- Studies on fate and effects of antifouling coatings released from wave and current energy devices, particularly as new products are developed.
- More sophisticated methods for assessing cumulative impacts.

Fishery Resources

- Fishery habitat maps for regional, site-specific, and cumulative impact assessment.
- Comprehensive approach to address limited baseline data, spatially and temporally.
- Integration of benthic community studies with fish and bird studies for predator-prey interactions.
- Dose and responses for the types of sounds from construction and operation for representative fish species.
- A definitive study to answer whether EMF from cables directly affect sensitive fish species (attraction, avoidance, prey detection).
- Consequences of the introduction of artificial hard substrates on fishery sustainability, energy flow, and the fisheries communities.
- For ocean current devices, studies on the impacts of turbine speed, water flow, and water pressures on fishes.
- Models and methods to evaluate management of alternative energy parks as fishery resource enhancement areas.

Marine Mammals

- Baseline studies are needed for key species in each region. Studies need the following:
 - Exploration and development of platforms and methodologies for improved monitoring and assessment of marine species;
 - Dynamic models that link environmental conditions at the time of each sighting; and

- Appropriate sampling resolution and study duration to provide sufficient baseline data for impact detection.
- Impacts of pile-driving and operational sounds on marine mammals for different settings and species hearing capabilities, to define the zones of influence, determine the effects, and develop appropriate mitigation measures.
- Experimental studies on the response of marine mammals to increased vessel traffic (sound, disturbance, harassment).
- Studies to assess attraction, collision risks, and impacts on movement through ocean current facilities.

Sea Turtles

- Baseline studies in high-priority areas for species density, importance, specific uses, and routes in and around the area.
- Impacts of pile-driving and operational sounds and vessel traffic noise on sea turtles for different settings and different species hearing capabilities, to define the zones of influence, determine the effects, and develop appropriate mitigation measures.
- Studies to determine if sea turtles are attracted to offshore energy installations, by lights and increased prey, then evaluate the consequences.

Discussions opened with consideration of benthic mapping needs. Although more information is desirable, some data are available for many areas. It was noted that the baseline studies funded by MMS at OCS sand and gravel sites would be applicable. Benthic habitat mapping was considered to be an area where MMS could work with other agencies to fill an important information need. There are several benthic habitat initiatives that would be good collaborative efforts. However, consistency in protocols, classification systems, and products is essential to produce edge-matched maps. Benthic habitat maps would be an important data layer when planning for further studies of the use of these habitats by key aquatic resources.

There was strong support among group members for regional studies that would provide baseline data for impact assessment; however, as important, was the need to ensure siting of initial alternative energy development in areas to avoid significant potential impacts and conflicts. This approach was used in the UK during Round 2 where a Strategic Environmental Assessment (SEA) was used to identify offshore areas for alternative energy development. MMS indicated that identification of priority areas for regional studies would be driven by the level of industry interest.

The group identified the need to understand how alternative energy facilities might affect vertical diurnal movements and on/offshore migrations. Most of the assessments have discussed alongshore migration patterns; however, there are major offshore-estuarine migrations that should be considered, including anadromous fishes (and protected species, in particular).

The group identified the importance of long-term monitoring studies to understand natural variations in distribution, abundance, and the influence of other factors (particularly climate change) for key species of concern. It was noted that many studies of offshore wind parks in Europe and the UK showed no significant differences at the wind parks compared to reference

sites because of large temporal and spatial variations in abundance. Group members suggested guidance is needed on which species and habitats should be included in baseline studies and monitoring. Information on threatened and endangered species is needed for consultations with other Federal agencies. It was suggested that other species can be monitored as surrogate indicators for broader groups with similar life history and behavior.

The issues of multiple uses of offshore alternative energy sites and the potential for exclusion of fishing were considered to be very important. There are many complex biological and socioeconomic issues with both de facto exclusion areas and specifically designated exclusion areas. Development is not likely to occur in highly productive fisheries areas (to avoid impacts and conflicts), thus any designation of a wind park as a marine protected area would do little to reduce direct fishing pressure.

Based on the discussion and earlier voting, the group developed eight main categories of information needs organized around types of potential disturbances or impacts that could affect all aquatic species, as follows:

- EMF Attraction and avoidance
- Noise Human and animal
- Migratory Issues Local, long distance, vertical, and on/offshore
- Endangered Species
- Collision/Entanglement/Entrainment Fishes, marine mammals, and sea turtles
- Habitat Changes New substrate, biofouling community, range extension for invasive species, sediment alteration, scouring, and change in current flow and wave regime
- Attraction/Avoidance Lighting, structure, and prey concentration
- Contaminants Oil spills from increased vessel traffic and antifouling coatings

Each of these categories was recorded onto a large poster with columns for the five areas of information being sought by MMS (data needs, suggested methods, collaborators, other limitations, special considerations). The group posted written comments on sticky notes in these columns for each category. Tables 1-8 are summaries of the comments by the group for each category.

Table 1 is a summary of the comments provided on the topic of EMF attraction and avoidance. Information is needed to adequately assess the effects of EMF on sensitive fishes and benthic organisms, particularly protected and rare species. There was concern that the transmission cables from offshore facilities to shore would cause attraction or avoidance and result in habitat fragmentation effects. For species with small or highly localized populations, this could lead to a population-level effect.

Table 2 is a summary of the comments provided on the topic of noise impacts. More information is needed on sound frequencies, levels, and propagation in settings where alternative energy facilities will be sited in the U.S. There was concern that existing data were only for single units rather than fully operational arrays where sounds are generated from both operations and maintenance activities. The best data will come from well-designed monitoring programs at actual operating sites in the U.S.

1			1	
Special Considerations	Consider both standard industry practices with respect to electric cables and anticipated next generation technology		Limitations of field studies, so may need controlled experiments; It is important to consider alternating current vs. direct current differences	Do not reinvent the wheel; look at all the research currently being conducted in Europe that was mentioned at this workshop
Other Limitations				
Collaborators	European Agencies; Cardiff University - effectiveness of acoustic avoidance and engineering solutions	UK Collaborative Offshore Wind Research into the Environment (COWRIE) EMF studies; University of Cranfield; Centre for Marine and Coastal Studies, Ltd. (CMACS)	NOAA National Marine Fisheries Service (NMFS)	Operators
Suggested Methods			Telemetry studies and lab evaluation coupled with population studies; Include visual behavior of species response to a submarine cable both buried and anchored on surface	
Data Needs	Development of a model calibrated with field and laboratory measurements of EMF levels produced for variations in cable size, capacity, sediment type, burial depth, etc.	Lab studies on fishes and detection/response to EMF for selected key species and life stages (e.g., salmon smolt), including surrogates for sensitive species	Study of areas with existing subsea cables and reference sites to assess impacts from measured EMF and evaluate potential for fragmentation of populations (cables could be linear blockages to movements)	Compilation of existing studies/data review

TABLE 1. Summary of comments on information needs for EMF – attraction and avoidance effects on aquatic resources.

TABLE 2. Summary of comments on information needs for noise – human and animal effects on aquatic resources.

		Noise (Human, Animal)		
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Monitoring data on sound	Towed and static deployed	COWRIE has measured noise	Both passive and active	Sound propagation can vary by
trequencies and levels for	hydrophones and devices	at offshore wind projects in	echo effects in arrays;	location; also location conditions
entire arrays, not just	currenuy deployed;	UK; should build on their	May need to consider	can vary seasonally
individual structures, for	Monitoring via pop-up buoys;	experience; Noise	distance from shore and	(temperature, salinity, wave
construction, operation, and	Other long-term monitoring	feasurements at EMEC and	types of noise	climate), so may be complex
Species-specific andiograms	Video/direct observation in	NMFS.		These are nermanent stationary
operies-specific autility	v luco/ullect ouser valuan III	II C Marrie		t nese are permanent stauonary etriotirae (imfile moving noise
thrachold shifts for morina	budroubones to dominant	CLU. Navy, Should huild on the extension		suuctures (unine moving noise
mommals and see truthes	mommel memory to memory	PIDUIU DUIIU UII UIE EXICIISIVE		sources), unus urere are potennar immore to local communities
especially threatened and	nuise levels to a particular	research mu ure enects of acoustics on marine mammals		
endangered species	technology	and fishes		
Studies on masking of noise		Jeremy Nedwell –		Transmission of above-surface
for fishes, marine mammals,		Subacoustech Ltd.; Jonathan		noise into the aquatic
sea turtles - hearing and		Gordon - Sea Mammal		environment via mooring lines,
communication		Research Unit, University of		cables, towers, etc.
		St Andrews, UK; Dr. Barry		
		Shepard - RPS Group		
Lab studies on fish perception		NMFS		Comparison and noise density of
of sound from construction				existing man-made noise levels
and operation activities to				and frequencies
establish dose/response for				
different endpoints				
Field data on the impacts of	Video/direct observation with			
turbine vibration on seabed to	hydrophones to document			
flatfish and benthic species	fish response			
Effectiveness of scheduling,		COWRIE - Current research		
bubble curtains, other		on effectiveness of Acoustic		
mitigation measures to reduce		Deterrent Devices and		
pile driving noise		engineering solutions for		
		dureter primg		
Long-term effects and				
activities in area				

	Migratory Is	sues (local, long distanc	ce, vertical)	
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Better understanding of the inshore/offshore and north/south migration and movement patterns (by life stage) for pelagic fishes (e.g., King/Spanish Mackerel) and benthic fishes (e.g., Gag grouper)	Fishery-dependent data may be of some value; however, fishery independent monitoring programs are much better (e.g., south Atlantic region, southeast area Monitoring and Assessment Program and Mid-Atlantic Regional Marine Research Program)	NMFS; Fishery Management Councils		Life stage shifts in habitat use; Fish distribution and behavior at different water depths since underwater turbines can possibly operate at various depths
More detailed data on marine mammals migration routes		NMFS		Mammal distribution and behavior at different water depths since underwater turbines can possibly operate at various depths
Role of introduced hard substrates in enhancing movement of species, including invasive species				Studies completed in other countries
Better understanding of how installations might affect vertical migrators				At current depths, vertical migration is not a prominent feature of system; there are some day/night migrators
Migration can be energetically expensive; can minor alternations lead to fitness consequences?				

TABLE 3. Summary of comments on information needs for migratory issues (local, long distance, vertical) on aquatic resources.

	Threate	ined and Endangered S	pecies	
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Baseline data on spatial and temporal distribution and habitat use in priority development areas for salmon, sturgeon, sea turtles, marine mammals	Current methods (aerial, vessel visual surveys) and improved techniques; acoustic tagging of fishes and sea turtles, passive acoustic hydrophones and T-PODS for mammals; Need improved methods to include environmental conditions and dynamic models on abundance and distribution data	NOAA Office of Response and Restoration – Environmental Sensitivity Index digital data have compiled existing datasets; NMFS; NMFS; NMFS; NMFS; NMFS; Program; Fishery Management Councils; USFWS		Consider areas targeted for reintroductions of extirpated species (e.g., sea otters along the Oregon coast)
Characterizations of sites/areas that provide habitat to specific species	Research methods used by USFWS and NMFS endangered species programs	Same as above		Are there any benefits to these species due to exclusion zone protection?
Species-specific data on sensitivity to the various stressors (noise, contaminants, EMF)		European agencies		
Extent of exposure to potential impacts - collision/entanglement, contaminants, noise, habitat alteration, etc. from single projects and cumulatively (energy and other activities in an area)		European agencies	Cumulative impacts will be difficult to evaluate until actual monitoring data on impacts are available	Particularly concerned about wild salmon vulnerability to EMF and predation in wave arrays; It will be difficult to understand how impacts from noise, contaminants, etc. have consequences on the survival or reproduction of individuals and then on populations
Evaluate the potential to design installations so the site could be beneficial to some species (i.e., refuge)				

TABLE 4. Summary of comments on information needs for aquatic threatened and endangered species.

Col	llision/Entanglement/Er	ntrainment (fishes, marir	ne mammals, sea tui	rtles)
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Approach to quantify and evaluate increased risk of collision from construction and service vessel traffic, considering local and regional scales, for marine mammals and sea turtles	Historic data of collisions and entanglements in area; Modeling risk of collision/entanglement for single project and for several projects in a region	NMFS; USFWS		Primarily a rare, threatened, and endangered species concern
Effective measures to reduce risks of service vessel collision: avoidance protocols and techniques		NMFS; USFWS		Continuous vs. periodic monitoring; Can periodic monitoring be conducted to effectively predict collisions at all times?
Information on which to estimate probabilities for entrapment/entanglement between structures (mostly wave and current technologies), particularly for listed species	High resolution sonar and camera systems mounted on devices; currently being tested/proposed for open hydro and lunar energy in the UK; Environmental audits of remote facilities	UK researchers/developers that are testing devices (e.g., Lunar Energy, Ocean Power Delivery, Open Hydro)		Primarily a rare, threatened, and endangered species concern
Designs to reduce potential entrapment/entanglement Analysis of the species, life stage, and significance of entrainment in wave/ocean		Engineering and design experience from hydro power plants should be researched Research and assessments in support of liquefied natural gas facilities		Consider designs to reduce attraction to reduce risk

TABLE 5. Summary of comments on information needs for collision, entanglement, and entrainment for aquatic resources.

TABLE 6. Summary of comments on information needs for habitat changes for aquatic resources.

		Habitat Changes		
(new substrate, biof	ouling community, sedi	ment alteration, scourin	ig, change in current	flow and wave regime)
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Better habitat maps on which to evaluate potential impacts	Examine MMS Marine Minerals Program sand and gravel studies for baseline and potential for habitat change	Consider adopting NOAA Coastal Services Center marine habitat classification scheme for habitat mapping		
How to evaluate consequence of introduction of hard substrate in soft sediments (artificial reef effect) – changes in benthic community, predatory effects, fish community, spread of invasive species, etc.	Review studies of existing installations, e.g., recently published report by University of Plymouth Marine Laboratory (UK) that reviewed all UK projects		Assessing cumulative impacts will require detailed information of degree of habitat change per project	Increase in local habitat heterogeneity typically leads to increased trophic complexity and higher energy flows; Benefits from cessation of bottom trawling; Climate change may also be a factor in shifts in species distributions
How to evaluate consequence of a change in habitat from pelagic to fixed reef community on structures in the photic zone	Consider National Science Foundation, Long Term Ecological Research project for wind farm as fixed reef community	National Science Foundation		Need to consider secondary and tertiary food web consequences
What are the changes in food availability (low trophic level) at offshore wind parks	Ecological sampling, observations			
Potential impact to eggs and larvae using the Gulf Stream	NOAA/commercial wave/current/sound transport models			
Improved and standardized approaches and protocols for monitoring benthic community and effects	Before After Control Impact (BACI) approach for species abundance changes; Review studies/results from monitoring of oil and gas platforms			

	Attraction/Avoidanc	ce (lighting, structure, pi	rey concentration)	
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
How to reduce attraction due to lighting	Research alternative designs for navigation lighting (color, masking, etc.)	Commercial light designers		Lighting can attract prey that will attract species of concern, with increased risk of entanglement/entrainment
What is the potential for areas with large arrays to serve as local refuges for certain species	Direct observation; Hydroacoustics	NMFS; USFWS; Fishery Management Councils		Numerous data are available for single structures, artificial reefs, however scale of arrays could be the important issue
Do structures attract/concentrate prey and predators to small area; also attraction of recreational fisherman to reef	New methodologies for impact assessment to fisheries on ecosystem basis			

TABLE 7. Summary of comments on information needs for attraction and avoidance impacts for aquatic resources.

TABLE 8. Summary of comments on information needs for impacts of contaminants on aquatic resources.

There are insufficient data on the audiograms for key species of marine mammals, in particular, so it is difficult to assess responses, both lethal and behavioral, for the different types of sounds generated from offshore energy installations. Impacts to benthic resources, including flatfishes, from noise and vibration were also discussed. There are concerns that the sounds from alternative energy installations could mask the sounds used by animals for communication, prey identification, and other behaviors. It was acknowledged that pile-driving sounds would be of particular concern, thus more information is needed on appropriate mitigation measures to reduce impacts during the construction phase.

Table 3 is a summary of the comments provided on the topic of migratory issues for aquatic resources. There was concern that offshore installations could become barriers to both along shore and on/offshore migration patterns for marine mammals, sea turtles, and fishes. Barriers to estuarine migration were a newly identified concern. Vertical migration patterns by plankton were also of concern. The introduction of hard substrates in otherwise soft-substrate habitats could provide stepping stones for range expansion of invasive species as well as species of concern. It was acknowledged that studies would also have to consider other factors affecting range expansions, such as climate change.

Table 4 is a summary of the comments provided on special concerns for threatened and endangered species. Most of the concerns are also addressed in the other topics, but the group felt that it was important to specifically call out concerns for these protected species. MMS will have to consult with NOAA and USFWS, so they will need species-specific information on abundance, use, and potential impacts, both direct and cumulative.

Table 5 is a summary of the comments provided on collision, entanglement, and entrainment concerns. Collisions from increased vessel traffic (construction and maintenance) may be of concern in marine mammal and sea turtle concentration areas for all types of technologies. Methods are needed for evaluating the increased collision risks. Entanglement and entrainment were more of a concern for wave and ocean current technologies, where the devices are anchored or tethered to the bottom and have open and/or moving parts. Developers need to be informed of the importance of designing devices and projects that minimize these risks, particularly when threatened or endangered species may be affected.

Table 6 is a summary of the comments provided on habitat change concerns. It was acknowledged that offshore alternative energy installations are likely to change the habitat, which will lead to changes in communities, species interactions, energy flow, etc. Guidance is needed on how to determine the significance of these changes (e.g., major, minor, insignificant). Cumulative impacts could be significant as projects expand or new projects are built adjacent to existing facilities.

Table 7 is a summary of the comments provided on attraction and avoidance concerns resulting from lighting, structure, and prey concentration. Navigation lighting should be designed to minimize the risk of attraction of prey, which could attract predators of concern, such as marine mammals, sea turtles, and pelagic fishes. Introduction of large arrays of hard structures may attract certain species and serve as a local refuge. Again, the scale of large arrays, compared to a single oil and gas platform, requires more study.

Table 8 is a summary of the comments provided on contaminant concerns. The group was concerned with the increased risk of oil spills from the increased service vessel traffic at the offshore installations, where containment and recovery is very difficult. Oils used in the devices should be selected so they have the least environmental risk, even though the volume of oil in individual devices is relatively small. Wave and ocean current developers will need to use safe technologies to address marine fouling that do not release toxicants into the water. Once these antifouling technologies are identified, studies will be needed to determine their concentrations and fate in water, sediments, and food pathways.

Active collaboration among government agencies, industry, and academic researchers is essential. Collaboration should start with the design of devices to minimize environmental impacts. There should be more collaboration among agencies on regional baseline studies. Industry will have to be involved in site-specific monitoring. However, because the initial postconstruction monitoring programs will generate key data on actual impacts that will inform future projects, they will need to be well designed and have strong collaboration among stakeholders to give credibility to the results.

During the plenary session, the Aquatic Resources group reported on their discussions. The key discussion topics were identified as:

- Invasive species
- Estuarine migrations
- Marine Protected Areas (exclusion zones)
- Footprint of structures relative to the environment
- Collaboration with industry and agencies doing similar research in the European Union
- Competition for use on OCS similarities of impacts from different activities

When asked what the "show stoppers" were, the key issues were identified as potential impacts to threatened and endangered species, especially in migration areas, and changes to fish habitat. For the species of concern, it will be necessary to compile and evaluate the existing information on distribution and abundance and identify information needs, then conduct the field studies to fill those needs and support the assessment of potential impacts and monitoring. Furthermore, research is needed to develop platforms and methodologies for improved monitoring and assessment of marine species in offshore areas.

4.0 Flying Animals Breakout Group

The Flying Animals breakout group covered the following resources: birds, bats, and flying insects. There were 22 members in the group with representation from Federal agencies, academia, industry, non-governmental organizations, and international experts. The first session started with the following summary of the information needs identified in the Synthesis Report:

- Baseline data on distribution and abundance by species group for on-water and in-air movements and behavior during migration, wintering, foraging, and staging to assess the risk to bird populations. Specific studies for key species include:
 - Define flyways: distance from shore, density within migratory corridors, and timing of spring and fall migration.
 - Determine flight height including understanding inter- and intraspecies variability
 - Understand diurnal and nocturnal movements
 - Understand foraging and wintering use of offshore habitats
 - Define potential use of offshore areas by threatened and endangered species
 - Identify temporal patterns
 - Improve baseline data on distribution and abundance of all species groups in offshore habitats
 - Analyze the effects of weather on migratory and other movements
- Monitoring protocols for using acoustic, radar, and thermal imaging for attraction and avoidance behavior studies.
- Better tools for evaluating mortality at offshore wind parks: acoustic, accelerometer, and thermal target detection systems.
- Improved predictive models related to collision and displacement (with new data).
- Understanding the extent and context to which bats fly offshore: migration patterns, corridors, weather impacts, flight altitude and characteristics, group size, feeding behavior, and temporal variation.

These needs were based on a literature review of past studies on onshore wind farms, European offshore wind parks, and modeling results. The major focus was wind turbines, which is the only technology that has been thoroughly investigated. The group then discussed the needs for further research for flying animals, with a clear emphasis on offshore wind parks.

There was strong support for regional studies to be completed and used as the basis for siting of projects and the design of project-specific studies. The priority areas for regional studies were identified as the mid-Atlantic, southern New England, and the Gulf of Maine, based on where projects are likely to be built in the near future and bird/bat resource concerns. There was much discussion on study design and products. Baseline studies should not be based on migration patterns, because setting a fixed migration period or pathway can leave out important species and temporal considerations. Studies should not identify flyways, since the flyway can dramatically change based on species, weather, and inter-annual climatic changes. Instead, studies should map abundances by species for different periods and be integrated with weather conditions. Flight height is an important parameter, as is nocturnal movement. Thus, new methods and research technologies, such as thermal and acoustic detection systems, will have to be developed and tested on fixed platforms.

Because avoidance and attraction behavior of birds and bats are key concerns, monitoring studies at specific projects in the U.S. are needed to quantify these impacts and identify appropriate mitigation measures. It was acknowledged that the results of the European monitoring studies may not be applicable to the U.S. because of differences in species, behavior, site conditions, etc. However, development of tools to evaluate collision and mortality rates should be done in collaboration with researchers in Europe and the UK who have been working to develop such systems. MMS will need actual data to improve and validate predictive models for assessment of collision and displacement at future projects. Baseline assessment methods that need further development include acoustic, radar, and thermal detection systems.

To reduce risks to flying animals, more information is needed on the types and effectiveness of different mitigation measures. Studies were recommended at existing sites to evaluate different lighting techniques. It was noted that the Communication Tower Working Group (chaired by USFWS) has conducted research on lighting techniques for onshore towers and turbines; studies should be conducted first on land where it is easier to test and evaluate different approaches. Initial studies should include redundancy of methods to evaluate effectiveness of each method. For example, studies should use acoustic, accelerometer, and thermal imaging methods, simultaneously. Initial studies should be at inland wind projects. Once the effectiveness of methods tested inland has been evaluated, the effective components can be moved to offshore sites for further testing.

There is very little information on which to make even initial assessments of the potential impacts to bats. It is not known if they fly offshore, how far, or how high. There was discussion as to whether bats would be attracted to offshore towers. Survey methods for birds should be evaluated as to whether they can be used to detect bats as well.

There is a need for a risk assessment framework to evaluate the significance of results, particularly acceptable risk, since some impact is possible.

Based on the discussion, the group developed seven main categories of information needs for assessment of potential impacts to flying animals, as follows:

- Regional summaries for known abundance data for priority offshore areas Identify priority 3-D spatial/temporal distributions for both birds/bats
- Regional baseline studies in priority areas to fill data gaps for both birds/bats
- Synthesis of old/new flying animal data into geospatial abundance layers (by species/group to inform siting)
- Site-specific study protocols/guidelines for:
 - Collision/mortality detection
 - Attraction/avoidance behavior/change
- Improved predictive models that are based on new data and results
- Mitigation measures effectiveness evaluation at specific future installations (e.g., testing of different lighting configurations)
- Risk assessment framework to address key issues

Each of these categories was recorded onto a large poster with columns for the five areas of information being sought by MMS (data needs, suggested methods, collaborators, other limitations, special considerations). The group posted written comments on sticky notes in these columns for each category. Tables 9-16 are summaries of the comments by the group for each category.

Tables 9-11 show summaries of the comments provided on the topics dealing with creating good baseline data layers on the distribution, abundance, seasonality, and habitat use for birds and bats in priority offshore areas, where alternative energy development is most likely to occur. The first step (Table 9) is to compile and evaluate the existing data, much of which is quite old and patchy. Key data gaps should be identified, and recommendations made for new studies to fill these gaps. Data for shoal areas were mentioned as an important gap in existing NOAA data collected from ships that avoided these shallow areas. The USFWS has been working on data compilation, quality review, and digitization, thus they would be a good collaborator. Nisbet and Veit (in press) have prepared a summary of the status of all seabirds on the U.S. east coast.

New regional studies would then be conducted (Table 10). There are many requirements for new study methods and technologies because of the need for survey data at night, for flight height, during adverse weather conditions, by species, etc. New technologies need to be developed, tested, and validated.

With the new data and geospatial data analyses, improved geographic information systems (GIS) data should be generated (Table 11) for key species and species groups. These regional datasets are needed to support better siting analyses. They would also help identify the types of site-specific studies needed for individual projects.

Tables 12 and 13 are summaries of the comments provided on the need for site-specific study protocols and guidelines for a) collision and mortality detection, and b) attraction, avoidance, and behavior change. The group agreed that accurate assessment of avian mortality was critical for public confidence in the benefits of offshore wind energy. There are many difficulties in this work because there are no proven technologies to detect collisions and quantify mortality. Once effective methods being tested on land-based towers and turbines are developed, offshore platforms would be good test sites for further refinement for marine applications. The methods used need to provide definitive data on the actual mortality rates of offshore wind parks. There was discussion on the potential use of beached bird carcass surveys as one method to detect large-scale mortalities at offshore sites.

Study methods and technologies are needed for studies to determine attraction, avoidance, and other changes in behavior of birds in the vicinity of offshore wind parks. The studies in Europe and the UK are of limited value because the conditions and species in the U.S. are very different. There are very little data on attraction, which could increase risks of collision mortality for species normally not at risk.

y offshore	
priority	
data foi	
undance	
ן/bat abו	
s of bird	
ımmarie	
rional su	
s for reg	
on need	
nformati	
nts on ir	
commei	
nary of	
9. Sum	areas
TABLE	

hore Areas	Special Considerations	Distinguish between peer- reviewed studies and gray literature; For each study provide summary of how data were collected and discuss level of uncertainty and accuracy; Create searchable database by location or species; Look beyond ship-based data, e.g., historical NEXRAD data should be considered for high altitude birds	Before and after strong meteorological (or other) environmental events; Include all birds that pass through, not just seabirds	
ata for Priority Offs	Other Limitations	Existing data are old and very patchy (for the most part)		
ו Bird/Bat Abundance D	Collaborators	Data for mid and north Atlantic being compiled by USFWS; Mass Audubon's Nantucket Sound studies on terns, wintering sea ducks, etc.; Nisbet and Veit, 2007: Summary of status of all seabirds on U.S. east coast; Manomet Bird Observatory data from ~1975-1985 mostly Georges Bank and Gulf of Maine; National Audubon Society, Regional Ornithological Society; The Nature Conservancy Marine Planning Group	USFWS; University researchers	Bat conservation NGOs; University researchers
I Summaries for Knowr	Suggested Methods	Compile and evaluate all historic and current survey data for all OCS areas (not just prime wind areas); Identify data gaps for priority areas	Review existing methods and prepare matrix of methods and metrics	
Regiona	Data Needs	Summaries of species- specific (seabirds; shore birds; neotropic passerines; raptors; other water-related birds) data (temporal and spatial abundance) based on geographic area (use or reason for presence, such as breeding, staging, over- wintering, migration corridor)	New methodologies for new surveys	Compilation of what is known about offshore abundance, distribution, behavior of bats

	LS.
	/Da
-	ras
-	5
٢	Ю
	tps
	b0
1	late
Ξ	
۲	
1	1S C
	lle
	<u>S</u>
•	
•	Ĩd
	II
	les
	na
1	esl
	Î
	ase
-	I D
	ona
•	En la
	LL
٢	
	ea
	ne
	lon
1	nat
	Elo
	IUI
1	on
+	SIL
	ner
	ິ
1	5
	lar,
	UU
ζ	Sui
¢	5
ŗ	- -
	211
	A
E	-

Regi	onal Baseline Studie:	s in Priority Areas to Fill	I Data Gaps for Birds a	nd Bats
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Improved baseline data on	Test long-range detectors	USFWS;	Annual variations may	Baseline studies should be multi-
distribution and abundance	for imaging from shore,	U.S. Geological Survey	make mapping of large	year to consider temporal
of all species groups in	e.g., radar (NEXRAD) &	(USGS);	spatial distribution not	variations;
offshore habitats;	IR laser;	NOAA;	predictable;	Integrate with climatology;
Include surveys of shoal	At-sea radar;	University researchers;	Annual variation must be	Trajectory of migration
areas, which have significant	Include flight altitude,	National Audubon Society	considered in designing	important (leap frog vs. more
data gaps and are priority	nocturnal movements, and		before and after studies, so	direct long distance);
development areas;	weather variation;		important to understand the	These data need to document
Shoreline bird inventory	Boats (chartered and ships		statistical power of the data,	"background" changes against
during staging periods	of opportunity from		including altitude	which to gage changes thought
	NOAA) and planes should		measurements	to be due to wind farms
	be used			
Understand habitat use;	Develop sampling	USFWS;		It is now possible to get data on
Need assessments of food	protocols with USFWS	NOAA;		food distribution (ADCP);
resources and bird resources	and NGOs	University researchers		Correlate sea surface
and species variability				temperature with distribution
through annual cycles				abundance for patchy species
Baseline data on beached	Existing protocols carcass	NGOs		Scavenging behaviors should be
bird rates	census			considered
Extent and context of	Methods development	USFWS;		
offshore distribution of bats	needed	University researchers		

TABLE 11. Summary of comments on information needs for synthesis of old and new flying animal data into GIS.

roup to inform siting)	Special Considerations	Provide a data layer that	illustrates percent uncertainty	depending on study design;	Access to large datasets may be	difficult			
Layers (by species/gi	Other Limitations	Data record too short to	determine shifts in	population distribution,	abundance, prey availability;	Limited weather data for	past studies;	Inconsistencies in data on	sex, age, activity, etc.
ata into GIS Abundance	Collaborators	USFWS;	National Renewable Energy	Laboratory (NREL) avian	database is a good example of	synthesis of peer-reviewed	studies		
/New Flying Animal Da	Suggested Methods	Use CAMRIS (Computer	Aided Mapping and	Resource Inventory	System)				
Synthesis of Old	Data Needs	Spatial and temporal data	on species distribution,	abundance, and habitat use					

TABLE 12. Summary of comments on information needs for site-specific study protocols and guidelines: Collision/mortality detection.

Detection	ns Special Considerations	 Will need to partner with industry to allow blades/towers to be modified; Use models to figure out what are the most important data to collect (sensitivity analysis) 	Learn from European studies; Develop guidelines on when each method should be used and what the results will show; le Need to develop a method for adverse weather studies ler		Protocols should be developed with USFWS and NGO
Ilision/Mortality I	Other Limitatio	No proven technologic so will take time to develop	Difficult to validate percent detection of different methods at se where collection of carcasses is not possib because of rapid predation, sinking, oth losses of fallen birds		
ols and Guidelines: Co	Collaborators	California Energy Commission; NREL: USFWS; NOAA – National Weather Service, NMFS; National Audubon Society; Communication Tower Working Group	Communication Tower Working Group; European researchers; NREL experience in developing protocols for land-based wind	Industry; Universities; USFWS	USFWS; NGOs
e-Specific Study Protoco	Suggested Methods	Build platform for collection of data on attraction and collisions - semi-permanently established at wind park site; Need radar/observation platforms for sites (movable)	Initial studies should include redundancy of methods to evaluate effectiveness of each (acoustic, accelerometer, and thermal imaging); Ensure that any studies using these technologies are coordinated with similar effectiveness studies being undertaken for land-based wind development	Identify scavenge rates during seasonal changes, for species- specific rates, considering sea state, time of day; regional variation; Observe bird events (movement and strike mortality) and document weather before/after the event	Validate use of carcass drift with telemetry to evaluate potential large kill with weather
Sit	Data Needs	Accurate assessment of avian mortality which is critical for public confidence in offshore wind energy	Methods to detect collisions remotely	Determine scavenging rates based on known (observed) consumption and rate (over time); this will allow more accurate estimates of mortality when assessing mortality events	Evaluate use of beached bird surveys for monitoring mortality rates

TABLE 13. Summary of comments on information needs for site-specific study protocols and guidelines: Attraction/Avoidance.

į

Data NeedsSuggested MethoMonitoring protocols for using acoustic, radar, and thermal imaging for attraction and avoidance behavior studies; behavior studies;Build on experience in Europe and UK; Need to test both visual remote methods; NEXRAD may be relati inexpensive tool to mea species to study (of concern, surrogate)				
Monitoring protocols for using acoustic, radar, and thermal imaging for attraction and avoidance behavior studies;Build on experience in Europe and UK; Need to test both visual remote methods; NEXRAD may be relati inexpensive tool to mea species to study (of concern, surrogate)Monitoring protocols for thermal imaging for attraction and avoidance behavior studies;Build on experience in Europe and UK; Need to test both visual remote methods; NEXRAD may be relati inexpensive tool to mea whiomass aloft" over lon term pre-construction	Methods Collabo	orators	Other Limitations	Special Considerations
using acoustic, radar, and thermal imaging for attraction and avoidance behavior studies; NEXRAD may be relati NEXRAD may be relati NEXRAD may be relati inexpensive tool to meat species to study (of concern, surrogate) term pre-construction periods	nce in USFWS;		Flocks may not be co-	Collect data on collisions during
thermal imaging for attraction and avoidance behavior studies;Need to test both visual remote methods;attraction and avoidance behavior studies;NEXRAD may be relati inexpensive tool to meas inexpensive tool to meas species to study (of concern, surrogate)surrogate)"biomass aloft" over lon term pre-construction periods	NMFS;		incident with the presence	fog, rain, and night;
attraction and avoidance remote methods; behavior studies; NEXRAD may be relati Need to identify appropriate respecies to study (of concern, "biomass aloft" over lon surrogate) term pre-construction periods	I visual and Universities		of migrating prey at a	Integrate with studies on prey
behavior studies;NEXRAD may be relatiNeed to identify appropriateinexpensive tool to measSpecies to study (of concern, surrogate)"biomass aloft" over lon term pre-construction			specific location	availability and change;
Need to identify appropriate inexpensive tool to mea: species to study (of concern, "biomass aloft" over lon surrogate) term pre-construction periods	be relatively			A lot of avoidance data - need to
species to study (of concern, "biomass aloft" over lon surrogate) term pre-construction periods	to measure			focus on attraction;
surrogate) term pre-construction periods	over long			First projects in each region
periods	ction			serve as prominent platforms for
				monitoring equipment
Guidelines for assessment of Review results of UK	fUK		Make sure that long-term	Can a standard protocol and data
impacts, for specific project Cumulative Impacts	acts		effects are considered, and	requirements be developed for
and cumulative Workshop (May 07)	07)		that there is not bias	carrying out a cumulative impact
			against certain species	assessment?

TABLE 14. Summary of comments on information needs for improved predictive models.

lts	s Special Considerations	 Other effects – Weather, prey changes, abundance variations, etc. How effective are the existing predictive models? Are there data to evaluate this? Can specific models be used for specific situations? Include measures of uncertainty for baseline and mortality data in models 	What are the likely impacts of displacement (food availability/energy expenditure)? Are there models to predict this?
lew Data and Resul	Other Limitations	Probability will differ by species, so will need species-specific study da Land-based efforts have been unsuccessful in developing a predictive model (see California Energy Commission stud	
<i>r</i> e Models – Based on N	Collaborators	National Wind Coordinating Committee (NWCC); Universities; USFWS	USFWS
Improved Predictiv	Suggested Methods	Time varying probabilities based on statistical distribution, weighted averages, rotor swept areas, etc.; Consider cumulative impacts from all stressors	
	Data Needs	Improved predictive models related to collision and displacement; Habitat suitability models for offshore habitats, including water depth, seafloor substrate, prey distribution/seasonality, etc.	Guidelines for how to evaluate consequences of collision and displacement

	Mitigation Measure	es Evaluation at Specific	c Future Installations	
Data Needs	Suggested Methods	Collaborators	Other Limitations	Special Considerations
Develop a list of potential mitigation measures acceptable to both industry and regulators	Review mitigation methods employed elsewhere for effectiveness, including both on land and offshore; UK monitoring data will start to show whether the mitigation measures in licenses have been/are being effective; Findings of independent peer-review monitoring panel	Industry; NGOs; USFWS; Electric Power Research Institute (EPRI) – Presently they are developing information on avian vision; Communication Tower Working Group	Compare mitigation strategies to other energy/OCS projects (e.g., sand and gravel) for fairness issues	A structure for integrating mitigation measures and changing restrictions (shut down times, etc.) must be built into the regulations and permitting process; Monitor effectiveness of mitigation and revise accordingly (e.g., follow adaptive management principles)
Develop rulemaking with conservation measures that will mitigate known (predicted, verified) impacts to wildlife resources	Habitat loss/displacement compensation buyouts, based on magnitude of loss	NOAA; The Nature Conservancy		Use royalties for habitat/compensation banking/restrictions
Avian light attraction data: aviation and navigation obstruction lighting	Controlled experiments with aviation and navigation obstruction lighting offshore			Review existing studies for power and wind projects on land – lighting; MMS may have access to offshore installations where lighting research could occur

TABLE 15. Summary of comments on information needs for mitigation measures evaluation.

TABLE 16. Summary of comments on information needs for a risk assessment framework to address key issues.

	Risk Assessm	nent Framework to Addr	ess Key Issues.	
ЭМ	ethods	Collaborators	Other Limitations	Special Considerations
Use existing	frameworks for	Develop joint guidance	Need to involve experts	Annual (or seasonal) iterative
ouner energy Measure pote	projects; intial bird	assessments with	and proad range of stakeholders in	input of population statistics from decade-fixed points;
mortality aga	inst other	collaboration among external	development of risk	Should be for specific key
known adver	se	consultants, USFWS, and	framework – Review	species;
environment	al impacts from	MMS	lessons learned from	Focus on the decisions to be
climate chang	ge, fishing		NREL and land-based	made for alternative energy on
practices, etc			experiences in developing	the OCS;
Standardize c	output		risk framework	Standardize treatment of
probabilistica	lly rather than			uncertainty for output, recognize
deterministica	ully			uncertainty in analytical
				techniques deployed, how data
				were collected, and how data
				were analyzed
Explore use of	expert	National Academies;		Would help determine what will
elicitation as a	technique to	Risk experts involved with		happen at a specific site
reduce site-spe	cific	"structured decision making"		
uncertainties ir	avian studies			
and the compar assessment	nion risk			
				What is an acceptable level of
				impact on a population? (As
				industry expands, have to expect
				at least some imnact)

Table 14 is a summary of the comments provided on the need for improved predictive models once actual data on abundances, behavior, collision, and mortality risks are available. There may be a need for specific models for specific conditions. These models will be used to assess impacts at future sites. They should include measures of uncertainty. With improved models, guidelines will be needed for how to evaluate the output, in terms of significance of predicted impacts. Eventually, it will be important to consider cumulative impacts from all stressors.

Table 15 is a summary of the comments provided on the need for evaluating and testing effective mitigation measures. The group discussed two types of mitigation measures: 1) measures to be implemented as part of the design or operation of a facility (e.g., shut down of turbines during periods of peak bird migration); and 2) compensatory measures to offset predicted impacts to resources (e.g., creation of additional nesting habitat for the number of endangered roseate terns estimated to be killed by turbines).

Table 16 is a summary of the comments provided on the need for a risk assessment framework to address the key issues associated with impacts to flying animals.

The results of the Flying Animals breakout group were presented at the Plenary Session. Each of the priority information needs was discussed, and the results of voting on these topics were presented. The need for regional baseline studies in priority areas to fill data gaps in the knowledge of the density and habitat use of offshore areas for birds and bats was ranked as being of highest priority. The need for site-specific study protocols and guidelines for detection of collision and mortality was ranked second, and detection of attraction and avoidance behavior was ranked third. The need for testing and evaluation of mitigation measures at specific future installations (e.g., testing of different lighting configurations) was also of priority. During the discussion, it was reiterated that more consultation with other agencies and experts was needed; thus, the group recommended that the first steps needed were a data summary and data gap analysis.

5.0 Physical Oceanography and Air Quality Breakout Group

The Physical Oceanography and Air Quality breakout group covered the following resources and processes: waves, currents, sediment transport, and air quality. There were 18 members in the group with representation from Federal agencies, academia, and industry. There was little discussion on air quality, probably due to the fact that few of the participants at the session were meteorologists. The first session started with the following summary of the information needs identified in the Synthesis Report:

- A method or approach for establishing acceptable limits to the impacted environment caused by modified physical processes.
- Additional monitoring of waves and currents at existing facilities: offshore of the development, within the development, and in the lee.
- Shoreline morphology monitoring.
- An accurate way of representing alternative energy structures in numerical models for farfield modeling.
- Data sharing infrastructure should be established, and any data sharing requirements should be specified.
- Determining the extreme farfield impacts of extracting a significant amount of energy from an ocean current.
- Assessment of the ability of these structures to impact the stability of the complete ridge or shoal upon which they are founded.

The initial discussion was on the need for an assessment that would lead to generation of maps identifying where the wind, wave, and ocean current resources were suitable to support alternative energy extraction, and how these areas spatially relate to electrical demand and grid capacity. Existing maps of energy potential are based on limited data. It was agreed that national-scale mapping of renewable resources potential was a key information need, to attract developers, to prioritize areas for future assessment, and to support environmental assessments.

Sediment transport issues included potential changes within the development site (nearfield) and along adjacent shorelines (farfield). It will be important to know the natural rates of change so that impacts from the development can be detected and quantified. Sediment dynamics need to be studied over the long term (decadal scale) to form adequate baseline data. Because many of the early development sites may be located on sand shoals, where the water depths are shallow, the sediment transport processes and rates need to be better understood. The MMS Marine Minerals Program is conducting studies of ridge and swale features that are important borrow sites for shoreline restoration projects along the east and Gulf of Mexico coasts. These studies will contribute to the understanding of the sediment dynamics of these features.

The issue of potential impacts to sediment transport focused on the lack of criteria to determine what is an acceptable level of change. Models can be used to predict shoreline changes, for example; however, without knowledge of the natural variation at a given site, it is difficult to assess the significance of the effects of the offshore structure. Within a project site, changes to sediment transport patterns may be important in terms of local scour around foundations; however, changes in grain size can have biological implications in terms of benthic communities,
habitat use, and trophic transfer. Similarly, participants felt a need to make sure that impacts to the physical processes are connected to further impacts to biological and conflicting-use resources.

Two types of air-related impacts were discussed. Increased vessel traffic during construction, maintenance, and decommissioning could be a local source of air pollution. However, the benefits of alternative energy (reduced combustion of fossil fuels for power generation) would greatly offset these impacts. There was some concern and uncertainty about the effects of wind parks on air velocity and turbulence, thus affecting local climate and air quality onshore, in the lee of wind parks. It was decided to limit the discussion on air quality as the participants felt that impacts on air quality due to alternative energy development on the OCS would be significantly less than impacts to other physical processes.

There were similar farfield concerns expressed with ocean current energy, particularly on the Gulf Stream, where farfield impacts could include nutrient mixing and potentially modified micro-climates. Farfield concerns with respect to wave energy extraction included shoreline and bank morphology.

Analyses of potential long-term consequences are complicated by predictions that wind, wave, and current fields may be significantly modified by climate change, and some participants suggested including future climate change scenarios in long-term analyses.

Many of the participants expressed concern about the cumulative impacts of multiple projects in areas with the highest energy potential. Examples include ocean current extraction in the Gulf Stream, wave extraction in high-energy wave climates on the west coast, and on wind energy extraction in the high-energy wind regions of the Northeast.

Based on the discussions, the group identified the following main themes for information needs:

- Need for geospatial tools Resource mapping (wind, ocean current, waves), biological resource mapping, onshore energy demand and grid connection
- Baseline of existing conditions Nearfield
- Baseline of existing conditions Farfield
- What are impacts from structures Nearfield, including stability of shoals and ridges
- What are impacts from structures Farfield, including impacts of loss of energy from a system
- What to do with quantitative impacts results Need some guidelines on what levels of change are acceptable

Each of these categories was recorded onto a large poster with columns for the five areas of information being sought by MMS (data needs, suggested methods, collaborators, other limitations, special considerations). The group posted written comments on sticky notes in these columns for each category. Tables 17-22 are summaries of the comments by the group for each category.

processes IIII	part assessingues.			
		Geospatial Data Tools		
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Alternative energy resource maps for offshore areas – Combined modeling and	Coupled modeling and observation studies: 1. Collection of existing data;	NOAA; U.S. Army Corps of Engineers (USACE) for	Density of measurements; Encompassing all	Shared funding by many agencies necessary because of great cost
observational data for wind, wave, and ocean current	 Preliminary modeling to define spatial scales, regions of uncertainty, and identify 	nearshore waves; Navy; States;	relevant processes (modeling); Resolving spatial scale)
	data gaps; 3. Field data collection; 4. New data used to	Contractors, private industry; Academia;	and resolution issues of the baseline data	
	refine models	Private sector should be encouraged (incentives) to share data		
Maps of demand in relation		Utilities;		Expected growth in electricity
to resource and grue capacity		Department of Ducigy		uennanu, Regularly updated "living database"
Geospatial tool with all	Interdisciplinary studies to	Academia (all areas dealing		
natural biological resources,	look for models	with science issues in		
utilities, effet gy definatios, nhvsical processes etc for				
siting analysis and impact				
assessment				
Approaches to determine				
biological consequences of				
much destratification or				
frontal creation or decrease is				
acceptable Internation with regional and	Evamlae.			
other related interest groups	-CODAR deployments near			
together with their data and	Cape Hatteras and New Jersey			
monitoring networks.	-Regional wave models -NDBC hilov data			
	-prior science studies, e.g. SEEP, OMR, PRIMER			

TABLE 17. Summary of comments on information needs for geospatial data tools to support energy resource mapping and physical processes impact assessments

•))	
Baseline	of Existing Conditions (waves, currents, wind,	sediment transport)	- Nearfield
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Better understanding of morphology and sediment	Field data collection and numerical modeling	USGS; NOAA;		Consider ongoing studies of shoals - sediment transport
transport patterns and rates on		USACE;		processes and rates - by MMS
shoals, to be able to		Consultants;		Marine Minerals Program
predict/detect changes post		Developers		
construction				
Long-term metocean data	In-situ monitoring,	Regional Associations/IOOS;	Little historical data, so	Integration of site monitoring
(wind, waves and currents)	CTD/ADCP;	NOAA	long-term datasets are	into regional Ocean Observing
farther offshore;	Remote sensing;		not possible yet – will	Systems - ability to use
Baseline monitoring –	Numerical modeling;		need to use hindcasting	modeling capabilities and tap
stratification and frontal	Hindcasting		and forecasting methods	into physical and biological
structures				monitoring of arrays in place
TADIF 10 Commence	· · · · · · · · · · · · · · · · · · ·		-t - t - t - t - t - t - t - t - t - t	

TABLE 18. Summary of comments on information needs for developing a baseline of existing conditions in the nearfield.

TABLE 19. Summary of comments on information needs for developing a baseline of existing conditions in the farfield.

arfield	Other Limitations Special Considerations	eed to identify where Temporal issues (decadal vs. focus initial efforts to short term) llect data – where are ilkely sites	Historical shoreline change rates can be determined from aerial photography; NOAA charts may provide data for offshore changes		
f Existing Conditions – F	Collaborators	States; N USGS; to USACE; co NOAA Coastal Services Ctr; th Universities	NOAA (buoy and bathymetric data); MMS (sediment studies)	NOAA; Regional Ocean Monitoring Networks	USGS; USACE;
Baseline of	Methods	Field sampling using LIDAR; In-situ testing; Numerical modeling; Hindcasting; Current change	Analysis of historical aerial photography/hydrographic charts; Shoreline change monitoring programs	In-situ testing (buoys or masts); Numerical modeling; Hindcasting	HF Radar; Sediment transport models
	Data Needs	Data and approaches to predict impacts to shoreline sediment transport mechanisms and other forces driving geomorphic change	Historical sediment transport patterns for analysis of changes after offshore development	Wind, waves and currents further offshore of present datasets	Understanding deepwater sediment transport processes;

TABLE 20. Summary of comments on information needs for assessing the nearfield impacts of structures.

	Impac	ts of Structures – Near	field	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Data and models to predict and mitigate local and global scour near foundations in conditions similar to the U.S. OCS.	Pre- and postconstruction multibeam side scan sonar mapping and bathymetry; Numerical models	Developers with government oversight; Researchers at existing projects in Europe and UK	No actual monitoring data for U.S. settings for model validation and refinement	Processes are sub-meter in scale
Data and improved models to predict changes in currents, waves, and sediment transport within the project footprint; Effects of spacing and water depth so can minimize impacts through project siting/design	Field studies at existing sites with in situ instrumentation for currents, wave, wind using HF radar, ADCPs, etc.; Protocols for postconstruction monitoring of projects in the U.S.; New model development	Developers with government oversight; Researchers at existing projects in Europe and UK	No actual monitoring data available for U.S. settings for model validation and refinement; Scaling – prototype testing over time	New approaches to physically represent the turbine bases in models – spacing is too small for grid size; Long-term data for impacts to waves, wind, and currents are sparse; Need real field data to improve and validate models; May need to consider changes in fishing practices (exclusion of trawl fisheries) that could affect sediment dynamics
Methods to assess potential for stratification and alteration to mixing of nutrients in the euphotic zone	Field CTD; Buoys; 3D Hydrodynamic modeling; MODIS Ocean color	National Space Administration Agency; Academics; Private		

TABLE 21. Summary of	comments on information n	eeds for assessing the farf	ield impacts of structures	-i
	Impac	ts of Structures – Far	ield	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Models to predict shoreline changes	Sediment transport models driven by site-specific data on wave climate, bathymetry, etc.	Developers with government oversight	Depositional environment shoreline type	Defining significant change compared to natural variation, coastal development, climate change, sea-level rise, etc.
Data to assess impacts to wave/current fields, sediment transport in lee of facilities	Monitoring of existing site; Model refinement	Developers with government oversight	May need nearby undeveloped site for reference	Models need improvements to generate the wind field in lee more accurately
Does energy extraction alter stability of the Gulf Stream?			Climate change impacts on Gulf Stream	Meandering of Gulf Stream brings nutrients onto shelf – changes could have large, cumulative impacts
TABLE 22. Summary of	comments on information n	eeds for how to apply qua	ntitative impacts data on	physical processes.
	How to ap	ply quantitative impac	ts data?	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Wave height reduction from development on the OCS, what is an acceptable impact?	Shoreline deposition changes; Public meetings	USGS; State agencies; Local interest groups	Shoreline type and use would influence acceptable tolerances	It will be important to include extreme weather events in the analysis
Current, waves, sediment concentrations, tides	Statistical analysis; Database development	Academia	Data format and consistency - data sharing	Seasonal variations require long-term data collection
Social data on acceptable impacts	Create new government- managed group	USEPA; Academia; Industry; Public		Determining regulatory compliance
National database	Interactive GIS database – internet enabled mapping system (resource limiting conditions)	NOAA- NODC		Temporal issues – long term vs. medium term vs. short term
Better understanding of cumulative impacts of multiple installations; how to define the acceptable density		NOAA; Developers; Industry; Universities		

Table 17 is a summary of the comments provided on the needs for geospatial data tools to support energy resource mapping and physical processes impact assessments. The group wanted more detailed analysis and maps of the energy potential in the OCS to support policy development, attract developers, assess impacts to physical processes, and identify linkages and interactions with biological resources. They hoped these maps or spatial databases would include metocean data/climates, seafloor properties, space-use conflict areas, electrical demand and grid capacity, as well as sensitive biological areas. This effort would require compilation and synthesis of existing data and collection of new data to fill data gaps, which they thought would be significant. They felt the priority should be first collecting the existing datasets and establishing links with regional monitoring networks and other agencies collecting data (e.g., NOAA).

Table 18 is a summary of the comments provided on the needs for developing a baseline of existing conditions in the nearfield around potential development sites in the OCS. Emphasis was on shoals and ridges where wind energy developments are most likely to be located in the near term because of their shallow depths. Research being funded by the MMS Marine Minerals Program on sediment transport and impacts of dredging at offshore sand borrow sites should be directly applicable.

Table 19 is a summary of the comments provided on the needs for developing a baseline of existing conditions in the farfield of potential development sites in the OCS. Emphasis was on baseline (historical) data and understanding of nearshore sediment transport processes on which to evaluate potential changes postconstruction of offshore projects. Where historical metocean data are unavailable, hindcast data were seen as a passable (although less-preferred) substitute. Historical data on shoreline change are available or may be determined using aerial photography, but not on shoreline sediment transport patterns and rates within the entire littoral zone, or historical data on their driving forces (wind, waves, currents offshore out in the OCS); this will need to be connected using numerical models. In the lee of likely development locations in the OCS, more detailed monitoring of shoreline change and sediment transport was also thought prudent.

Table 20 is a summary of the comments provided on the needs for assessing the nearfield impacts of structures. The group discussed three types of impacts: 1) local scour around foundations; 2) broader changes in waves and currents influencing sediment transport processes and thus sediment characteristics in the footprint of the development site; and 3) broader changes in water currents and circulation that could affect water quality inside the development site. The results of monitoring studies at existing wind parks in Europe are of only limited value because conditions at U.S. sites are very different. Model refinement is needed to address issues of the fine scale needed for analysis, for the actual size and spacing of structures as well as the physical processes to be modeled.

Table 21 is a summary of the comments provided on the needs for assessing the farfield impacts of structures. Three areas of concern were discussed: 1) shoreline changes from offshore wind and wave energy developments; 2) changes in wave and current field and thus sedimentation patterns in the lee of developments (between the project site and the shoreline); and 3) impacts of

current energy extraction on the Gulf Stream. Existing models need refinement and validation based on site-specific monitoring studies.

Table 22 is a summary of the comments provided on how to apply quantitative impacts data on physical processes to support impact assessments. Modelers can generate data quantifying changes in terms of a percent reduction in wave height, or a distance over which currents are reduced. However, there is no guidance on what degree of change should be considered acceptable, for both direct and cumulative impacts. In the desktop studies conducted in Europe, the impacts to physical processes were always considered insignificant.

The group also discussed the importance of information sharing during all phases of a project and encouraged MMS to develop an information-sharing program.

Overall, the physical oceanography and air quality group concluded that, at the present time, efforts should be focused on the marine-based impacts. They believed that concerns with the air quality impacts were either less significant, or less likely to occur. The most significant need identified by the group was the need for bringing data into one place-preferably a spatially referenced database/mapping tool. The group did not limit what should be included, but specifically identified winds, waves, currents, seabed characteristics and stratigraphy, energy resource potential, electricity demand and grid capacity, space-use conflicts (navigation, fishing grounds, etc.), biological indicators, and sensitive areas, as well as numerical modeling results. It is believed that this system would provide an efficient means for determining where to focus the efforts of further data collection and modeling. It would also provide a framework for project developers and MMS to collaborate on effective siting; all participants agreed proper siting was the primary and most effective impact mitigation approach. The group encouraged the collaboration among regional monitoring networks and other agencies (NOAA, USGS, etc.) to populate the database and identified expanding regional networks and other existing programs as perhaps the most efficient means of filling physical data gaps and establishing baseline monitoring.

Nearfield impacts of the developments focused on the better understanding of the physical processes going on within the footprint of developments, and some felt it would require the development of new approaches to accurately represent the structures in a numerical model. Farfield impact discussions were dominated by impacts to shorelines; however, the farfield impacts of a modified Gulf Stream were identified. The group generally felt that any farfield impact assessments were limited by the fact that at present, wind, wave, current, and nutrient datasets do not extend far enough offshore to be useful to all alternative energy applications on the OCS. A discussion surrounding all of the impacts included concerns with the cumulative impacts of numerous installations, and how MMS will determine the acceptable limits of the cumulative impacts. Linkages between modified physical processes and biological impacts were not discussed in detail; however, it was felt that a good initial approach at establishing those linkages was to include biological indicators in the spatial database.

6.0 Social Sciences and Economics

The Social Sciences and Economics breakout group covered a broad range of topics related to feasibility of projects, policy, regulations, permitting, and socioeconomic impacts. There were 21 members in the group with representation from Federal agencies, academia, industry, and the international community. The first session started with the following summary of the information needs identified in the Synthesis Report and by MMS staff:

- Conduct of studies on policy, socioeconomic impact
- Detailed guidelines for visual impact assessments
- Determination of thresholds of visual impact
- Impacts to onshore land use
- Impacts to onshore infrastructure
- Economic impacts job creation or income generated
- Impacts to indigenous tribes and subsistence fishing
- Effects on recreation and tourism
- Impacts to navigation and transportation impacts to space
- Communications impacts to radar

Breakout group members focused much of the initial discussion on economic issues of project development. MMS stated that it is not the government's role to determine if a proposed project is economically viable; that is a business decision. However, impacts in different areas are within the government's scope. Group members suggested that array size can affect both economic and environmental impacts. The number of arrays will affect cumulative impacts. Economic viability will affect scenarios selected for assessing impacts. And, economic viability will affect the developer's willingness to take on monitoring and mitigation requirements. It was also pointed out that the size and location of the lease area may determine what a developer realistically can and can not propose.

The economic viability discussion led the group to the subject of cost/benefit analysis. Group members suggested that cost/benefit analyses of alternative energy projects should be holistic and broad, address issues such as differences between renewable sources and fossil fuels which are finite, and consider the subsidies provided to conventional fuels, regional differences in energy demand versus sources, environmental and human health effects associated with air pollutants, climate change, and experimental technologies, and how some issues such as visual impacts may vary by geographic area. It was acknowledged that costs and benefits should be determined regionally, not nationally, because of significant regional differences, even though renewable energy is in the national interest. National policies are important; in the UK, there is a strategic national policy that drives renewable energy project development. The group concluded that guidelines on cost/benefit analysis are needed.

A subset of the cost/benefit analysis discussion focused on the appropriate methodology for assessing traditional socioeconomic impacts such as those on tourism, property values, or fisheries as an input to the cost side of the cost benefit equation. Surveys and case studies are two different approaches to those types of assessments that have their specific advantages and disadvantages. Neither approach is standardized. Surveys are sometimes used to assess local acceptance of alternative development options, but for the local people, the results are often viewed as unreliable. There are a variety of concerns with surveys ranging from lack of trust in the survey sponsor to the mechanics of the survey itself – either the form or content of the questions or the sampling method.

The issue of experimental vs. commercial projects was raised a number of times in the discussion of economic viability and cost/benefit analysis. It was suggested that experimental projects should be treated differently than commercial projects, in terms of the leasing process, including any cost/benefit analysis, and permitting requirements, since they can not be expected to be economically viable. Deployment of meteorological masts should fall under the leasing process of experimental projects. The question was raised as to whether the environmental assessment for leases for experimental projects, i.e., those testing technology, would/should be different than the assessment for commercial projects.

As was the case for all the breakout groups, there were repeated requests for maps showing areas of alternative energy resources overlain on maps showing energy needs, infrastructure, environmental and use conflicts, and other concerns. This kind of information is essential for initial project scoping, siting, and economic analyses of costs and benefits. Industry is very interested in MMS providing geographical data and tools for initial project evaluation. Maps at the regional level were specifically requested in order to make sure that geographic and economic nuances are captured. Group members suggested a number of collaborate efforts that they felt would enhance any mapping effort and improve the quality of information available at the regional level. These included working with the shipping industry, commercial fisheries groups, fishery management councils, research tied to the mapping effort, and existing resources from Federal, State, and local sources.

Throughout the discussion, the group grappled with understanding the differences between the established MMS oil and gas program and the still under development alternative energy program.

The group's discussion fell into the following major categories of issues and concerns:

- Alternative energy adaptive management (as applied to monitoring and mitigation requirements)
- Commercial vs. experimental technologies
- Cost/benefit methodology
- Space-use issues and conflicts
- Understanding stakeholders
- Mapping
- Regional and national issues—meta-level
- Regulatory coordination/collaboration

Each of these categories was recorded onto a large poster with columns for the five areas of information being sought by MMS (data needs, suggested methods, collaborators, other limitations, special considerations). The group posted written comments on sticky notes in these

columns for each category. Tables 23-29 are summaries of the comments by the group for each category.

Table 23 is a summary of the comments provided on the use of adaptive management principles as applied to monitoring and mitigation requirements. Because these are new and evolving technologies, there are many requirements for monitoring to determine the magnitude of potential impacts. These requirements should be appropriate to the scale of the project. The study results should be used to shrink or change future requirements to reflect the current understanding or knowledge base. Studies should not automatically be required for every new site. MMS should periodically review the results of past studies and revise monitoring requirements.

Table 24 is a summary of the comments provided on how the different phases of project development should be treated. The group thought that the experimental phase of a project, usually installation of instrumentation to collect site conditions data on wind, waves, currents, etc., should have minimal requirements for permitting and leasing. Because of the need for demonstration projects in U.S. settings, such projects should be supported, with different requirements. MMS should also support information sharing for all projects.

Table 25 is a summary of the comments provided on the needs for improved methodologies for cost/benefit analyses of alternative energy projects. This topic had the most number of "sticky note" comments for this breakout group. There was a clear need for more "hard" data on which to make these kinds of analyses. Also, there were many comments on quantifying the true costs of conventional energy production, considering the future costs of climate change and the broad consequences to humans and natural resources. Such holistic analyses will require both social and economic research to collect the necessary data and develop the appropriate methods of analysis.

Table 26 is a summary of the comments provided on space-use conflicts and mapping; these two categories were combined because there was overlap in the data, methods, etc. Mapping tools and the necessary data are needed for developers to identify and avoid space-use conflicts very early in the scoping process. During one of the presentations on the SEA process in the UK, it was clearly shown that simple overlays of uses are of limited value; nearly all areas show potential conflicts. Thus, more sophisticated methods are needed that allow ranking of factors and weighing of variables. With updated and higher-resolution maps of energy resources, developers will be able to identify potential sites within their areas of interest and get initial information on resource and conflict issues. Because there will always be some conflicts, developers in this emerging industry want guidance documents to help them collaborate with stakeholders so they can resolve conflicts early in the process. The guidance document should include information on possible mitigation measures to reduce or avoid conflicts. Developers need to benefit from the experience of others in this regard.

Table 27 is a summary of the comments provided on understanding stakeholders. Again, the group discussed the need for tools and documents to guide developers towards successful communication with stakeholders. Developers also voiced the need to better understand the MMS process for leasing and permitting. The group also identified the need for MMS to develop

Summary of comments on information needs for how to apply adaptive management principles to monitoring and mitigation requirements for alternative energy installations. TABLE 23.

7	Alternative Energy Adapt	tive Management (Mon	itoring and Mitigatio	(
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Guidance on how adaptive management principles should be applied to monitoring and mitigation requirements for alternative energy installations over time	Activating requirements in consent documents	NOAA; USFWS: Industry		Need to identify scale of project that would make certain monitoring worthwhile, required, and/or reasonable; Monitoring requirements for future projects should shrink or change to reflect the current understanding or knowledge base
Information needed to determine the minimum scale for assessing technology and impacts	Periodic review of current understanding of impacts	Natural resource agencies; NGOs		How do you balance encouraging investments in new technologies with the risk that future knowledge may find the impacts of these technologies unacceptable?

Data Needs	Treatment of Comme Methods	Collaborators	Other Limitations	Special Considerations
stinutions of technical (e.g., et towers), demonstration, d commercial projects, th demonstration bridging e gap	Definitions would include size, duration, experience level in the region, technology variations, and data to be acquired	Federal agencies; States; Industry; Utilities; Universities	Are criteria needed to assess the appropriate maximum scale of commercial projects?	Small-scale projects less than a specified amount of power generated are needed for operation, integration, and maintenance experience
assing and permitting quirements for different lases of projects, as listed ove atabase summaries of perimental technology formation, including trrent projects, study goals, rther links	Identify duration requirements; Fast track a ruling to allow data collection for R&D on the OCS (e.g., installation of met towers)	Federal agencies; States; Utilities; Universities Industry	In the U.S., demonstration projects are needed because there are concerns not addressed in Europe/UK projects, including hurricanes and lightning strike survivability; Regional variability in conditions, experience and knowledge of technology	Allow greater uncertainty/probability distributions for data requirements for experimental technologies used in demonstration projects; Less information would be needed for small-scale, time- limited R&D projects; Europe started with small projects that provided data and helped with public acceptance Should be a minimum requirement that data reports be posted to this new database by the project in order to get an experimental status (or noncompetitive status); Data sharing moves industry towards greater certainty and more narrow distributions as technology is demonstrated and moves to commercial status
&D funding		MMS; DOE		Federal funding moves projects along faster

Summary of comments on information needs for treatment of commercial versus experimental technologies. TABLE 24.

Summary of comments on information needs for developing cost/benefit methodologies for alternative energy projects. TABLE 25.

	Cos	t/Benefit Methodologie	S	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Better data on which to quantify benefits, such as	Surveys and interviews; Contingent valuation analysis;	Local business groups; Local and regional	Holistic approach is hard to agree on but needed;	Guidelines are needed on: how to value public policy
health benefits from lower pollutant emissions, reduced	Choice of appropriate discount rate:	stakeholders/interest groups; USEPA:	Monetary impact is very hard to quantify:	requirements in the analysis; How to quantify societal
greenhouse gases and climate	Determine levelized costs and	U.S. Department of Energy	Hard to understand full	benefits of renewable energy;
change effects on	benefits;		cost of non-renewable	Each region will need
environment and humans, oreater enerov indenence	Sensitivity analysis; Consider data from Furonean		generation alternatives considered in NFPA	different data and approaches; Need data on real costs of
and job gains	experience; Data for on land installations			conventional power
Better data on which to		I ocal business groups:	No hard data from prior	Production Fach region will need
durantify costs such as		Local and regional	experience.	different data and annroaches
installation and operating		stakeholders/interest groups;	Much is based on	Is there a conservation benefit
costs, grid integration,		USEPA;	speculation and personal	to exclusion zones/artificial
negative environmental		U.S. Department of Energy	drivers;	reef effects that should be
effects, negative effects on		Industry;	Every project will have	valuated vs. lost fishing access
property values, aesthetics,		Utilities	consultants that will work	costs?
fisheries, tourism, job losses			for and against the project	
Financial analyses of				Create an MMS royalty
alternative and conventional				structure that helps support a
energy – magnitude of				renewable energy industry on
subsidies; financial viability;				the OCS
potential for lease payments;				
subsidy lease payments				
Standards of methodology	Survey European experience;		It is time consuming to	Identify where standardization
should be consistent across	Workshops with key		make generic guidelines	of methodology is needed and
all similar industry sectors;	stakeholders;			where regional variability does
Consistency of language	Interviews with social			not warrant it;
usage across agencies	scientists and decision makers;			Recognize that methodologies
	Mapping of process to identify			will need to be adapted to suit
	areas needing standardization;			different technologies at
	BACI and dynamic methods			different stages of maturity
	where BACI does not work			

Cont.	
25.	
BLE	
TAJ	

	Cos	t/Benefit Methodologie	Sé	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Overarching examination of methods for understanding potential benefits and detrimental effects of emplacing offshore alternative energy facilities and development of standards of validity and reliability in their application	Literature review to identify relevant research methods and guidelines for application of social and economic research and analysis in natural resource use settings; Interview core experts and others who use information to identify desirable standards of reliability, validity, and precision; Existing information and regulatory regimes should be assessed for other similar industries (e.g., electrical power generation) creating a database of requirements to use as a reference; Convene inter-agency panels to clarify discrepancies and codify language	Fishery Management Councils; NMFS; Bureau of Land Management; National Park Service; State agencies with social science capacity; Key stakeholders; Developers	Involve state agencies to identify areas that could not be standardized and why; Vast number of stakeholders	Goal is to generate guidelines for the application of social and economic research methods in assessing potential benefits and challenges of alternative energy options on the OCS; Involve social science research specialists in project

Spi Methods Mifv GIS overlavs are simple bi	Collaborators	ping Other Limitations	Special Considerations
re simple bu fy potential where, thus ated methoc	it Shipping interests; Utilities; Military; Is State and Federal resource		Tools should be available to developers who can do their own analyses; Need to establish guidelines
include rar can be var	iks agencies; ied Fishery management		for buffers around certain types of conflicts (e.g.,
al conditi	ons councils; Local agencies; Trade associations; Commercial fishing groups		pipelines, shipping routes); Initial tool is needed soon, can refine over time
lata layers	t, Shipping interests;	Some data are	Should MMS compile all
uala, cval gaps,	Military;	Variations in data scale	developers or link to datasets
o collect	key State and federal resource	and detail may not	maintained by other agencies?
.0	agencies; Fisherv management	support project-level analvsis:	OMB authorization for surveys:
egion;	councils;	Will need the cooperation	Statistical grids used by
get	Local agencies;	of agencies who have the	fishery agencies too large
t (type, rioritiz	I rade associations;	data	resolution, needs to be refined through work with fisheries
7111011			and other users
vsis to	Shipping interests;		Generic mitigation strategies
ected	Utilities;		may not be applicable
s like	Military; State and federal resource		nationwide, however examples might speed resolution:
	agencies;		Education of locals: trips to
	Fishery management		offshore wind farms, meetings
	councils;		with all stakeholders including
	Local agencies; Trade associations:		fishing, tourist groups, NIMBV groups
	Commercial fishing groups		
nd, wa	ve,		If the NREL maps are used for
horem	läl		utistiote prospecting, freed
			more data on land effects on wind

Summary of comments on information needs for space-use conflicts and mapping. TABLE 26.

	Unde	erstanding Stakeholde	LS	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Guidance document for developers on how to communicate with stakeholders to promote local acceptance of projects and alternatives	Interview agencies and developers to identify needs and issues; Survey coastal communities and other stakeholders to identify key issues that may pose problems; Develop tools for productive outreach and communications; Develop ethnographic studies to document attitudes pre- and postconstruction	Federal, state, local permit entities; Community associations; Public officials; Tribal governments; Fisheries interest groups	Variations in regions and states	Must be clear to developers in terms of process; The different stakeholders in each region need to be defined; Get a small group of stakeholders together representing each region and brainstorm about the issues for each stakeholder group; Identify valid indicators for use in long-term socioeconomic monitoring framework
Building better partnerships between developers and MMS	Interview developers about their ideas on an accommodative "competitive" process, and demonstration projects that can be used to measure environmental impacts (to refine later generations of rules)	Industry	The expectation of studying everything before implementing anything needs to be removed	Developers need a defined permitting process to foster this new industry
Outreach program to provide the public with objective information on issues associated with alternative energy on the OCS	Develop methods that educate the general public, e.g., publish articles in local and national media; Develop polling methods on NIMBY issues in different regions and nationally		Need to realize that you will never satisfy everybody	Extend understanding beyond perceptual/attitudinal dimensions of offshore renewables to social and sociopolitical (behavioral) response to programs/projects

TABLE 27. Summary of comments on information needs for understanding stakeholders.

TABLE 28. Summary c	of comments on information	needs for regional and nati	ional issues.	
	Regi	onal and National Issu	es	
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Regional data on cost of power, utility reserves, energy needs, strength of transmission grid, variations in grid rules, competing priorities, etc.	Review of jurisdiction between Federal, State, and local agencies	Regional ISOs (independent system operations); Utilities		
Identify what issues are of national interest and which issues are of local interest; Identify most optimal way to address both	Surveys; Workshops with stakeholders; Review other agency experience and lessons learned, i.e., USEPA			
TABLE 29. Summary c	of comments on the need for	regulatory collaboration.		
	Re	gulatory Collaboration		
Data Needs	Methods	Collaborators	Other Limitations	Special Considerations
Create "one stop" for permitting; Establish a timeline for entire permitting process to enable developers to estimate expenses	Establish regional work groups with Federal, State regulators and grid operators to address cross jurisdictional process, consistency in regulatory regime	Federal agencies; State agencies; Developers; Public		Permitting needs to be driven by one department on the Federal level; Need to develop process for coordination with the States
Identification of issues in fishing, environmental, navigation, aviation; Data for each of these issues for strategic development areas around the country	Working groups on each issue on a national level, including industry reps and relevant regulatory representatives; Coordination by one agency	Federal agencies; State agencies; Developers; Public	Will not be able to capture all issues of local or regional importance or variability	Define parallel groups at the regional level to ensure two way transfer of pertinent information; Ensure emerging information is available to all
Regulatory guidance for developers - process flow map of regulatory/permitting procedure; A framework for	Meta-study that defines Federal, State and local regulators and jurisdiction for selected States	Federal agencies; State agencies; Developers; Public		May require regional analysis; Use of lean principals to optimize process while meeting requirements

procedure, A framework for coordination and clarity in regulations and schedules

a program of public outreach and education on the issues associated with alternative energy development on the OCS.

Table 28 is a summary of the comments provided on regional and national issues. Many of the comments overlapped significantly with previous issues. Table 29 includes comments on the need for regulatory collaboration. Representatives from this nascent industry voiced considerable frustration over the complexity, costs, and delays in permitting, leasing, and other regulatory compliance requirements. They encouraged MMS to provide for a "one-stop" process to assist developers. They requested guidance documents explaining the regulatory process and the issues that may arise during the process.

At the Plenary Session, the group presentation started with the fact that MMS's intent was not to determine economic viability of specific projects; however, there was significant discussion by the group of factors that would aid developers and others in understanding implications of critical parameters such as size of units, extent size of arrays, scope of lease areas, variation in environmental and socioeconomic factors across regions, etc.

The group identified a need to use lessons derived from renewable projects in Europe. They saw the need for and desirable content of comprehensive cost-benefit analyses that included consideration of the unique nature of renewables vis-à-vis traditional OCS resources.

The topics of most interest to the group were:

- Treatment of commercial vs. experimental technologies
- Space-use issues/conflicts
- Understanding stakeholders
- Standards of methodology
- Cost/benefit methodology

There were two key areas of concern. First, there was a clear need for development of tools for mapping energy resources and potential conflicts; stakeholder collaboration guidance; a guide for developers on permitting and leasing; and tools for improved communications between MMS and developers. Second, it was noted that the issues discussed by the group were not the traditional areas of study under the topics of social sciences and economics. Rather, there was extensive discussion on permitting and regulatory requirements. This was, in part, because the group thought that topics such as visual assessments had been adequately addressed in the synthesis report.

The group identified the following as examples of priority projects:

- Impacts of alternative energy projects on local tourism
- Space and use conflicts between commercial fishing and alternative energy projects
- Identification and perceptions of stakeholders
- Establishing/identifying state-of-the-art social science methodologies

7.0 Final Plenary Session

The session opened with a brief summary of the discussions over the first two days of the workshop. Then it was opened up to further open discussion, which is summarized below under the major information needs.

7.1 Geospatial Data and Tools

All four breakout groups identified the need for the compilation and evaluation of geospatial data and collection of new data to fill identified data gaps. Maps showing the locations of energy resources were of priority interest to every group: Utilities need them to assess the role that alternative energy may provide in future energy supplies; Developers need them to select the most feasible sites for further analysis; Regulators and resource managers need them to identify priority areas to start baseline data collection; and State and local agencies need them to inform the public on the issues. Higher-resolution, nationwide data are needed on energy resources, namely wind, waves, and currents. The existing maps need additional data and refinement of the analysis. Several groups are collecting data at specific sites. The U.S. Department of Energy will be an important collaborator.

Once the priority areas are identified for possible alternative energy development in the OCS, the Federal and State resource agencies will need to compile existing data on habitats of concern and the distribution, abundance, and use of these areas by species of concern to identify key data gaps and develop study plans to collect missing data to support regional assessments. Researchers noted that compilation of existing data requires extensive data validation and quality control before the data can be used in environmental assessments. Data compilations are also needed for other types of use conflicts, such as navigation, commercial and recreational fishing, tourism, cultural resources, etc.

There was discussion about what studies will be conducted by MMS and what studies industry will be asked to conduct. Though there are no hard rules, generally the MMS Environmental Studies Program will conduct or support studies that address generic issues (e.g., the effect of sound on marine mammals) or cover large, regional issues (e.g., regional circulation patterns). Industry would be expected to conduct studies at a specific site (e.g., benthic habitat mapping in the footprint of the installation).

There are many opportunities and a great need for collaboration and partnering to conduct the necessary mapping of natural resources. This mapping effort is needed to identify areas that should be avoided, support impact assessments of proposed projects, identify resources that may need more detailed study and/or mitigation to reduce impacts, and provide baseline for monitoring studies. This workshop was one step toward fostering partnerships among agencies to address the environmental issues associated with alternative energy development in the OCS. It will be important to continue collaboration.

7.2 Geospatial Analytical Tools

Each group also discussed the need for geospatial tools to support data analysis at different scales. Developers need to quickly identify areas to avoid, areas with potential conflicts or

issues, and areas with the least potential conflicts. They also need to identify the environmental and socio-economic resources that may need to be addressed during environmental assessments.

Specialized tools will be needed for data analysis and integration across disciplines. For example, weather is a very important factor influencing the distribution and abundance of birds offshore; thus, researchers need tools to help them acquire and analyze weather information in combination with bird survey data.

7.3 Evaluation of Mitigation Measures

The MMS Environmental Studies Program conducts studies on the effects of development activities in the OCS and supports research on strategies to avoid or reduce significant effects. This is one area where collaboration with other groups working on similar problems is essential, such as work being done by the Communication Tower Working Group on methods to detect and reduce bird strikes on wind turbines.

7.4 Development of Guidance Documents and Study Protocols for Impact Assessments

Standardization in field methods for data collection, classification, and assessment was another common theme among the breakout groups and in the plenary discussions. Some recommendations were specific to technical issues, such as methods for detection of collisions by birds with turbines, and new techniques to assess abundance of sea turtles in an area. Other recommendations focused on guidance documents for use by developers to communicate better with the public and to better understand permitting requirements.

7.5 Overarching Messages and Conclusion

During the plenary session discussions on the final day, the participants identified three areas needing MMS attention most immediately: 1) development of the Alternative Energy Program; 2) data collection and information management; and 3) collaboration and coordination. These areas represent the high-level results of the workshop and are areas where MMS efforts are already underway.

To support development of the Alternative Energy Program, participants indicated that a strategic assessment of regions where environmental information collection is needed would be valuable. The European experience and activities can serve as a guide for programmatic marine spatial planning. Stakeholders would benefit from guidance on the process and regulatory structure. Stakeholder involvement has been recognized as a key component of the Alternative Energy Program, and there are continuous efforts underway to ensure their participation.

Data collection and information management are of primary importance to the Environmental Studies Program (ESP). In this area, evaluation of the most important information needs is being undertaken currently and is supported by synthesis and consolidation of recent studies. Workshop participants frequently stated that the need for various map layers that contain current, high-quality data and that can be correctly aligned was essential. Coordination and collaboration with stakeholders and potential partners are key to the success of alternative energy studies. Tapping into existing resources will minimize duplication of effort, ensure that all concerns are addressed, and result in better scientific products. Drawing upon expertise in the international community will be important to continue, and that dialog will enhance study designs.

Also during the final day's plenary session, workshop participants provided their perceptions of the greatest challenges to the ESP for information collection. These can be categorized into three areas: 1) programmatic issues; 2) collaboration and coordination; and 3) methods and data. This discussion highlighted some known concerns already being addressed.

Programmatic issues challenging the ESP include the availability of staff, funding, and time. It takes all three to build and complete a long-term strategy for alternative energy studies. The challenge lies in developing sufficient useful information quickly enough to proceed with sound decision-making in the near-term.

The need to continue collaboration and coordination efforts was reiterated by the group. Building new relationships and enhancing existing ones will be a critical component in filling alternative energy study needs. Partnerships need to be fostered at all levels, from local, through State and Federal, to international communities.

Participants recognized both the existence of necessary data and the need for additional data. Challenges to the ESP when handling existing data include gathering and synthesizing it; ensuring the scale and scope of the data are comparable; and standardizing methodology. As important as avoiding duplication of efforts already completed or underway is recognizing when enough information has been collected. Prioritizing needs and leveraging partners through collaboration and coordination will ensure the best uses of staff, funding, and time resources.

The basic tenet of the ESP mandate is to provide the scientific information necessary to assess the impacts of offshore activities on the human, marine, and coastal environments and support environmentally sound decision-making. The results of this workshop will feed into the studies development process that provides scientific information for the Alternative Energy Program. The ideas and information generated at the workshop will be used to develop a cohesive and comprehensive study plan that will provide critical information to support programmatic decisions. The ESP will continue to communicate with stakeholders and potential partners identified and further foster the efforts initiated at this workshop.

Appendix A

Workshop to Identify Alternative Environmental Information Needs

Workshop Agenda

Minerals Management Service Alternative Energy Workshop Agenda

Day One – Tuesday, 26 June 2007

08:30 Registration

WORKSHOP INTRODUCTION

09:00 Elizabeth Burkhard, Workshop Leader Minerals Management Service, Herndon, Virginia Welcome, Workshop Purpose, Background, and Overview of Agenda

CURRENT STATE, FUTURE TRENDS AND LESSONS LEARNED Presentations will be delivered by experts covering the current state of wind and wave

- technologies, future trends, environmental concerns and lessons learned.
- 09:15 Kurt Thomsen Advanced Offshore Solutions, Arhus, Denmark Current state of and future trends in AE development in Denmark, factors influencing site selection, lessons learned
- 10:00 Steffen Nielsen Danish Energy Authority, Copenhagen, Denmark Environmental and social impacts of wind farm development in Denmark, lessons learned
- 10:45 BREAK
- 11:15 Robert Thresher National Renewable Energy Laboratory, Golden, Colorado Current state of and future trends in wind technology and associated environmental concerns in the US
- 12:00 LUNCH (on your own)
- 13:30 Michael Hay British Wind Energy Association, London, England Current state of and future trends in AE development in the UK, lessons learned
- 14:00 Rachael Mills Department of the Environment, Food and Rural Affairs, London, England AE development in the UK - government perspective, lessons learned
- 14:30 Chris Jenner RPS Group Plc, Surrey, England Environmental concerns associated with alternative energy development in the UK, lessons learned
- 15:00 BREAK
- 15:30 George Hagerman Virginia Tech, Advanced Research Institute, Arlington, Virginia Current state of and future trends in wave technology in the US

- 16:00 Greg McMurray Oregon Department of Land Conservation and Development, Salem, Oregon Environmental concerns associated with wave technology in the US
- 16:30 Elizabeth Burkhard General Comments on the Day
- 16:40 Adjourn for the day

Day Two – Wednesday, 27 June 2007

- 08:30 Registration
- 09:00 Maureen Bornholdt

Minerals Management Service, Herndon, Virginia Progress of MMS Regulation Development, Document Production, Scheduling, Governmental Framework

- 09:30 Stephen Bowler and Nicholas Jayjack Federal Energy Regulatory Commission, Washington, D.C. Highlights of Marine-based Studies to Support Wave, Current or Tidal Energy Development
- 09:55 Elizabeth Burkhard

General Comments and Charge to Breakout Groups

BREAKOUT SESSIONS

- 09:45 Concurrent Facilitated Breakout Groups
 - Physical Oceanography and Air Quality
 - Biological Oceanography
 - Social Sciences and Economics

Each group will be given a 15 minute overview of the relevant information from the Literature Synthesis to provide the framework for the group discussions of information needs. Participants in each group will discuss and develop a list of critical information needs that are scientifically feasible. Break times will be determined within each group.

- 11:30 A member of each breakout group will take 10 minutes to present his/her discipline's list of information needs to the whole group in plenary session.
- 12:00 LUNCH (on your own)

PLENARY DISCUSSION SESSION

13:30 During this facilitated discussion session, participants will identify overlaps and cross-connections between ideas generated in the different breakout groups to streamline the lists. Each participant will identify the 3-5 most critical priorities in the streamlined lists in their opinions.

- 14:30 Public comment period Those wishing to comment must sign up by 12:00 at the registration table. Comments will be heard in the registration order and will be time limited.
- 15:00 BREAK

BREAKOUT SESSIONS

- 15:30 Concurrent Facilitated Breakout Groups Participants will return to the breakout groups to flesh out the top 3-5 information needs in the discipline with brief generalized descriptions and objectives. Completed descriptions of information needs will be turned in to the facilitators at the end of the day.
- 16:30 Adjourn for the day

Day Three – Thursday, 28 June 2007

08:30 Registration

PLENARY DISCUSSION SESSION

- 09:00 Facilitated Plenary Discussion Session The group will discuss mutual interest in, and opportunities for, collaboration and partnership on studies.
- 10:30 BREAK
- 11:00 Public comment period Those wishing to comment must sign up by 09:00 at the registration table. Comments will be heard in the registration order and will be time limited.
- 11:30 The workshop will conclude with a review of the discussions and closing remarks.
- 12:00 Adjourn the meeting

Appendix B

Workshop to Identify Alternative Environmental Information Needs

List of Participants

James Adamson Marinus Power, LLC 440 Louisiana St., Suite 625 Houston, TX 77002 713-236-0037 j.adamson@marinuspower.com

Jennifer M. Arnold Penn State Berks 15526 North Nemo Court Bowie, MD 20716 301-390-3861 shirundo@aol.com

Timothy Baker DOI Solicitor's Office 1849 C Street, NW Washington, DC 20240 202-513-0821

Bruce Biwer Argonne National Laboratory 9700 South Cass Ave., EVS/900 Argonne, IL 60439 630-252-5761 bmbiwer@anl.gov

Thomas W. Bjerstedt MMS/GOMR 1201 Elmwood Blvd. New Orleans, LA 70123 504-736-5743 thomas.bjerstedt@mms.gov

Maureen Bornholdt MMS 381 Elden Street, MS-4041 Herndon, VA 20170-4871 703-787-1300 maureen.bornholdt@mms.gov Jim Ahlgrimm US Dept. of Energy Wind & Hydropower Technologies Program 1000 Independence Ave., SW Washington, DC 20585 202-586-9806 jim.ahlgrimm@ee.doe.gov

Larry P. Atkinson VA Coastal Energy Research Consortium Old Dominion University Norfolk, VA 23529 757-683-4926 latkinso@odu.edu

Melissa A. Batum MMS 381 Elden Street, MS 4040 Herndon, VA 20170 703-787-1646 melissa.batum@mms.gov

David A. Bizot NOAA National Marine Sanctuaries 1305 East-West Hwy. (N/ORM6) SSMC4, #11500 Silver Spring, MD 20910 301-713-7268 david.bizot@noaa.gov

Mary C. Boatman MMS 381 Elden Street Herndon, VA 20170 703-787-1662 mary.boatman@mms.gov

Stephen Bowler FERC 888 First Street, NE Washington, DC 20426 202-502-6861 stephen.bowler@ferc.gov Jeffrey Brandt TRC 650 Suffolk Street Lowell, MA 01854 978-744-6006 jbrandt@trcsolutions.com

William S. Bulpitt GA Tech Strategic Energy Institute 800 West Peachtree St., Suite 400E Atlanta, GE 30332 404-385-6939 bill.bulpitt@energy.gatech.edu

Alexis M. Castrovinci ICF Consulting 9300 Lee Highway Fairfax, VA 22031 703-934-3313 acastrovinci@icfi.com

James H. Churchill Woods Hold Oceanographic Institution Water Street Woods Hole, MA 02543 508-289-2807 jchurchill@whoi.edu

Christopher Clark Geo-Marine, Inc. 2201 K Ave., Suite A2 Plano, TX 75074 972-423-5480 cclark@geo-marine.com

Rodney E. Cluck MMS 381 Elden Street Herndon, VA 20170 703-787-1087 rodney.cluck@mms.gov

Kim F. Coffman MMS 381 Elden Street, MS-4010 Herndon, VA 20170-4871 703-787-1221 kim.coffman@mms.gov Stephen K. Brown NOAA/NMFS, Office of Science & Tech 1315 East West Highway Silver Spring, MD 20910 301-713-2363 stephen.k.brown@noaa.gov

Sean M. Casey ICF Consulting 9300 Lee Highway Fairfax, VA 22031 703-934-3222 scasey@icfi.com

Michael A .Champ ATRP Corp. 7000 Vagabon Drive Falls Church, VA 22042 703-946-6655 machamp@aol.com

Jim Cimato MMS 381 Elden StreetHerndon Virginia 20170-4817 703-787-1749 James.Cimato@mms.gov

Katherine Clayton Ecology & Environment Inc. 1700 North Moore St, Suite 1610 Arlington, VA 22209 703-522-6065 kclayton@ene.com

Louis "Coke" Coakley FPL Energy 700 Universe Blvd., JES/JB Juno Beach, FL 33408 561-691-7060 coke_coakley@fpl.com

Camille E. Coley Florida Atlantic University 777 Glades Rd., ADM 207 Boca Raton, FL 33431-0991 561-297-3461 ccoley@fau.edu James H. Collins Tetra Tech, Inc. 10306 Eaton Place, Suite 340 Fairfax, VA 22030 703-385-6000 jim.collins@tetratech-ffx.com

Bradford Davis Ocean Energy Institute PO Box 380007 Cambridge, MA 02238 925-286-0854 brad.davis@dartmouth.edu

Robert J. Diaz, Jr. Virginia Institute of Marine Science Virginia Institute of Marine Science P.O. Box 1436 Gloucester Point, Virginia 23062-1346

Wendy S. Early Research Planning, Inc. PO Box 329 Columbia, SC 29202 803-256-7322 wearly@researchplanning.com

William Evans Old Bird, Inc. 605 W. State St. Ithaca, NY 14850 (607) 272-1786 wrevans@clarityconnect.com

Peter Fippinger NOAA 1315 East West Hwy., SSMCS, Room 15640 Silver Spring, MD 20910 301-713-1622 peter.fippinger@noaa.gov

Doug Forsell US Fish & Wildlife Service 177 Admiral Cochrane Dr. Annapolis, MD 21401 410-573-4560 doug_forsell@fws.gov Rara Courtney Good Harbor Consulting 8 Walker Street Gloucester, MA 01930 978-317-3321 fcourt@cove.com

George Detweiler US Coast Guard 2100 2ns Street SW Washington, DC 20593 202-372-1566 george.h.detweiler@uscg.mil

Frederick R. Driscoll Florida Atlantic University 101 North Beach Rd. Dania Beach, FL 33004 954-924-7221 rdriscol@oe.fau.edu

Carolyn Elefant LOCE/OREC 1717 K Street, N.W. Ste. 600 Washington D.C. 20036 (202) 297-6100 loce@his.com

Colleen Finnegan MMS 381 Elden Street, Mailstop 4010 Herndon, VA 20170 703-787-1275 colleen.finnegan@mms.gov

Jeremy Firestone University of Delaware Colleg of Marine and Earth Studies Newark, DE 19716 302-831-0228 jf@udel.edu

Richard Fristik USDA - Rural Development 1400 Independence Ave., SW Washington DC 20250 202-720-5093 richard.fristik@wdc.usda.gov Norman Froomer MMS 381 Elden Street Herndon, VA 20170 703-787-1644 norman.froomer@mms.gov

Po Chi Fung NOAA 8510 16th Street, Suite 200 Silver Spring, MD 20910 626-776-6111 pochi.fung@gmail.com

Jennifer A. Ghiloni Tetra Tech, Inc. 133 Fedearl Street Boston, MA 02110 617-457-8421 jennifer.ghiloni@tteci.com

Edward Glazier Impact Assessment, Inc. 205 North Channel Ave., Suite B Wrightsville Beach, NC 28480 910-200-9650 eglazier@ec.rr.com

Peter Goldman PRG Consulting 610 Queen Street Alexandria, VA 22314 703-473-3495 prgconsulting@comcast.net

George Hagerman VA Tech Advanced Research Institute 4300 Wilson Blvd., Suite 750 Arlington, VA 22203 703-387-6030 hagerman@vt.edu

Jeff S. Hammond Ecology & Environment Inc. 368 Pleasant View Drive Lancaster, NY 14086 716-684-8060 jhammond@ene.com Michael Fry American Bird Conservancy 1731 Connecticut Ave., NW Washington, DC 20009 202-234-7181 mfry@abcbirds.org

John R. Gasper Argonne National Laboratory 955 L'Enfant Plaza, Suite 6000 Washington, DC 20024 202-488-2420 jgasper@anl.gov

Steve Gittings NOAA/National Marine Sanctuary SSMC-4, N/ORM62, Room 11642 Silver Spring, MD 20910 301-713-7274 steve.gittings@noaa.gov

Gary Goeke MMS/GOMR 1201 Elmwood Blvd. New Orleans, LA 70123 504-736-3233 gary.goeke@mms.gov

Scott Graves ICF Consulting 9300 Lee Highway Fairfax, VA 22031 703-218-2662 sgraves@icfi.com

Deb Hahn Association of Fish & Wildlife Agencies 444 North Capitol Street, Suite 725 Washington, DC 20001 202-624-8917 dhahn@fishwildlife.org

Kim Harb National Ocean Industries Association 1120 G Street, NW, Suite 900 Washington, DC 20005 202-737-0926 kim@noia.org Richard Harness EDM, International 4001 Automation Way Fort collins, CO 80525 970-204-4001 rharness@edmlink.com

John W. Hayse Argonne National Laboratory EVS Bldg. 900, 9700 South Cass Ave. Argonne, IL 60439 630-252-7949 hayse@anl.gov

Martin Heinze MMS 381 Elden Street Herndon, VA 20170 703-787-1010 martin.heinze@mms.gov

Maurice L. Hill MMS 770 Paseo Camarillo, Suite 222 Camrillo, CA 93010 805-389-7815 maurice.hill@mms.gov

Zakiya J. Hoyett NMFS, Office of Habitat Conservation 1315 East West Highway SSMC3 F/HC Silver Spring, MD 20910 301-713-4300 zakiya.hoyett@noaa.gov

Robin Jamail NOAA 1100 Wayne Ave. Suite 1225, Room 1236 Silver Spring, MD 20910 301-427-2436 robin.jamail@noaa.gov Michael Hay British Wind Energy Association Renewable Energy House One Aztec Row Berners Road London N1 0PW 44 20 7689 1960 michael@bwea.com

Erin Healy ICF Consulting 33 Hayden Ave. Lexington, MA 02421 781-676-4043 ehealy@icfi.com

Michael Hill Hill & Kehne, LLC 2300 Wisconsin Ave., NW, Suite 300 Washington, DC 20007 202-558-2100 mhill@hillkehne.com

Keely Hite MMS 381 Elden Street Herndon, VA 20170 703-787-1103 keely.hite@mms.gov

Mary Hallisey Hunt Georgia Tech, Energy Institute 800 West Peachtree St., Suite 400E Atlanta, GE 30332 404-385-3065 mary.hunt@energy.gatech.edu

Nick Jayjack Federal Energy Regulatory Commission 888 First Street, NE Washington, DC 20426 202-502-6073 nicholas.jayjack@ferc.gov Chris Jenner RPS Group Plc Goldsworth House, Denton Way, Working Surrey GU21 3LG United Kingdom +44(0) 1483 746523 jennerc@rpsgroup.com

Scott Johnston US Fish & Wildlife Service Div. of Migratory Birds 300 Westgate Center Dr. Hadley, MA 01035 413-253-8557 scott_johnston@fws.gov

Kerry Kehoe NOAA 1305 East-West Hwy. SSMC 4 Silver Spring, MD 20910 304-563-1151 kerry.kehoe@noaa.gov

John King Coastal Programs Division (N/ORM3), Office of Ocean and Coastal Resource Mgt. 1305 East-West Highway Silver Spring, Maryland, 20910 301-713-3155 john.king@noaa.gov

Timothy Konnert FERC 888 First Street, NE Washington, DC 20426 202-502-6359 timothy.konnert@ferc.gov

Jeff Krauss National Ocean Industries Association (NOIA) 1120 G. Street, NW, Suite 900 Washington, DC 20005 202-347-6900 jeff@noia.org Jeff Ji MMS 381 Elden Street, MS-4041 Herndon, VA 20170-4871 703-787-1145 jeff.ji@mms.gov

Jeff Kehne Hill & Kehne, LLC 2300 Wisconsin Ave., NW, Suite 300 Washington, DC 20007 202-558-2100 jkehne@hillkehne.com

Jim J.Kendall 381 Elden Street, MS-4041 Herndon, VA 20170-4871 MMS James.Kendall@mms.gov

Chris Koehler Geo-Marine 2201 K Avenue, Suite A2 Plano, TX 75074 972-423-5480 ckoehler@geo-marine.com

Mark L. Kosakowski Ocean Surveys, Inc. 91 Sheffield Street Old Saybrook, CT 06475 860-388-4631 mlk@oceansurveys.com

Elizabeth A. Kress Santee Cooper 1 Riverwood Drive PO Box 2946101 Moncks Corner, SC 29461 843-761-8000 eakress@santeecooper.com Michael Kujuwa Winergy, LLC 150 Motor Parkway, Suite 425 Hauppauge, NY 11788 631-434-9100 mike@winergyllc.com

Emily Lindow NOAA National MarineFisheries Service Office of the Assistant Administrator 1315 East-West Hwy., SSMC3 #14552 Silver Spring, MD 20910 301-713-2239 emily.lindow@noaa.gov

Helen A.Manich Current to Current Corp. 35 Corporate Drive, Suite 400 Burlington, MA 01803 571-334-3101 helen@manich.com

James F. McVeigh SENTECH, Inc. 7475 Wisconsin Ave., Suite 900 Bethesda, MD 20814 240-223-5528 jmcveigh@sentech.org

Ellen L. Mecray NOAA 1315 East West Hwy. Suite 11411 Silver Spring, MD 20910 301-734-1193 ellen.l.mecray@noaa.gov

Shelby Mendez NOAA 1315 East West Hwy., Room 15743 Silver Spring, MD 20910 301-713-1622 ext. 207 shelby.l.mendez@noaa.gov Robert P. LaBelle MMS 381 Elden Street, MS-4010 Herndon, VA 20170-4871 703-787-1700 robert.labelle@mms.gov

Rafael V. Lopez NOAA 1313 East West Hwy., #14112 Silver Spring, MD 20910 301-713-4300 ralph.lopez@noaa.gov

Andrew McGillis Baird & Associates 627 Lyons Lane, Suite 200 Oakville, Ontario, L6J 5Z7 905-845-5985 amcgillis@baird.com

Chris R. Meaney NMFS Office of Habitat Conservation 1315 East West Highway SSMC3 F/HC Silver Spring, MD 20910 301-713-4300 christopher.meaney@noaa.gov

Sam Medlock Hill & Kehne, LLC 2300 Wisconsin Ave., NW, Suite 300 Washington, DC 20007 202-558-2100 smedlock@hillkehne.com

Jacqueline Michel Research Planning, Inc. PO Box 328 Columbia, SC 29202 803-256-7322 jmichel@researchplanning.com Rachael Mills Dept. for Environment, Food & Rural Affairs Nobel House 17 Smith Square London, SW1P 3JR +44 207 270 1983 rachael.f.mills@defra.gsi.gov.uk

Steve Newell ABS 16855 Northchase Drive Houston, TX 77060 281-877-6848 snewell@eagle.org

Bill O'Beirne NOAA Ocean Service 1305 East-West Hwy. SSMC 4 Silver Spring, MD 20910 301-713-3155 ext. 160 bill.obeirne@noaa.gov

Craig Olmsted Cape Wind Associates 75 Arlington Street, Suite 704 Boston, MA 02116 617-904-3100 colmsted@capewind.org

Rachel Pachter Cape Wind Associates 75 Arlington Street, Suite 704 Boston, MA 02116 617-904-3100 rpachter@capewind.org

Doug Pfeister Bluewater Wind 22 Hudson Place Hoboken, NJ 07030 201-420-1195 doug@bluewater.com Michelle K.Nannen EEA, Inc. 1239 Rt. 25 A, Suite 1 Stony Brook, NY 11790 631-751-4600 mnannen@eeaconsultants.com

Steffen Nielsen Danish Energy Authority - Energy Supply Amaliegade 44 DK 1256 Copenhagen K 45 33926752 SRN@ENS.dk

Barry T. Obiol DOI - MMS 1202 Elmwood Blvd. New Orleans, LA 70123 504-736-2786 barry.obiol@mms.gov

Sean M.O'Neill Ocean Renewable Energy Coalition 12909 Scarlet Oak Drive Darnestown, MD 20878 301-869-3790 soneill@symmetrix.biz

Audra Parker Alliance to Protect Nantucket Sound 4 Barnstable Rd. Hyannis, MA 02601 508-280-9082 audra@saveoursound.org

Liz Philpot Southern Company - Alabama Power Co. 600 North 18th Street Birmingham, AL 35291 (205)-257-5315 efphilpo@southernco.com John Poulin ICF Consulting 1725 Eye Street, NW Suite 1000 Washington, DC 20007 202-862-1176 jpoulin@icfi.com

Roger Pugliese South Atlantic Fishery Management Council 4055 Faber Place Drive, Suite 201 Charleston, SC 29405 843-571-4366 roger.pugliese@safmc.net

Bonnie Ram Energetics, Inc. 9010 D St, SW, Suite 100 Washington, DC 20024 202-479-2748 bram@energetics.com

Russell Raymond Energetics, Inc. 901 D. Street, SW, Suite 100 Washington DC 22024 202-406-4111 rwraymond@energetics.com

Michael C.Reed SENTECH, Inc. 7475 Wisconsin Ave., Suite 900 Bethesda, MD 20814 443-415-8668 mreed@sentech.org

Nicholas Rigas Clemson University 386-2 College Ave. Clemson, SC 29674 864-656-2267 nrigas@clemson.edu

Ralph Mark Rouse MMS 1201 Elmwood Blvd. New Orleans, LA 70123 504-736-2594 mark.rouse@mms.gov James M. Price MMS 381 Elden Street, MS-4041 Herndon, VA 20170-4871 703-787-1641 pricej@mms.gov

Dennis Quaranta Winergy, LLC 150 Motor Parkway, Suite 425 Hauppauge, NY 11788 631-434-9100 dennis@winergyllc.com

Dennis E. Rankin USDA - Rural Development 1400 Independence Ave., SW Washington, DC 20250-1571 202-720-1953 dennis.rankin@wdc.usda.gov

Anne M. Readel NOAA, Oceanic & Atmospheric Research 1315 East West Hwy. SSMC3, 11409 Silver Spring, MD 20910 301-734-1131 anne.readel@noaa.gov

Ken Richardson ABS 16855 Northchase Drive Houston, TX 77060 281-877-6506 krichardson@eagle.org

Alescia Roberto USDA - Rural Development 1400 Independence Ave., SW Washington, DC 20250-1571 202-720-1414 alescia.roberto@wdc.usda.gov

Celeste H. Rueffert MMS 381 Elden Street Herndon, VA 20170-4871 703-787-1324 celeste.rueffert@mms.gov Leslie A. Safier ICF Consulting 9300 Lee Highway Fairfax, VA 22031 703-218-2697 Isafier@icfi.com

Amy R. Scholik National Marine Fisheries Service 1315 East West Hwy., SSMCS, Room 13605 Silver Spring, MD 20910 301-713-2322 amy.scholik@noaa.gov

Mark Sinclair Clean Energy Group 50 State Street Montpelier, Vermont 802-223-2554 sara@cleanupgroup.org

Gene Smith NOAA Undersea Research Program 1315 East-West Hwy. Silver Spring, MD 21401 301-734-1008 gene.smith@noaa.gov

Justin J. Sobol Florida Power and Light Universe Blvd. North Palm Beach, FL 33408 561-304-5259 justin_sobol@fpl.com

Jim Tabor MMS 381 Elden Street, MS 3412A Herndong, VA 20170 703-787-1125 jim.tabor@mms.gov Dana K. Savidge Skidaway Institute of Oceanography 10 Ocean Science Circle Savannah, GA 31411 912-598-3344 dsavidge@skio.usg.edu

Jeffrey L. Shenot NOAA 1315 East West Hwy. Silver Spring, MD 20910 301-713-0174 ext. 189 jeff.shenot@noaa.gov

Odin A. Smith NOAA 1305 East-West Hwy. Room 6415 Silver Spring, MD 20910 301-713-7392 odin.smith@noaa.gov

Jeffrey B. Smith Tetra Tech, Inc. 10306 Eaton Place, Suite 340 Fairfax, VA 22030 301-330-2880 jeff.smith@tetratech-ffx.com

Stephanie Strength USDA - Rural Development 1400 Independence Ave., SW Washington, DC 20250 202-297-4964 stephanie.strength@usda.gov

Steven Textoris MMS 43171 Parkers Ridge Drive Leesburg, VA 20176 202-208-3731 steven.textoris@mms.gov
Robert Thresher NREL - National Wind Tech Center 1617 Cole Blvd., MS3811 Golden, CO 80401 303-384-6922 robert_thresher@nrel.gov

Carleigh Trappe NOAA/NOS 1305 East-West Hwy. SSMC 4, N/ORM3 Silver Spring, MD 20910 301-563-1165 carleigh.trappe@noaa.gov

Jeffrey L. Underwood US Fish & Wildlife 300 Westgate Center Dr. Hadley, MA 01035 413-253-8408 jeff_underwood@fws.gov

Robert J. Varley Lockheed Martin Maritime Systems & Sensors 9500 Godwin Drive, B105/055 Manassas, VA 20169 703-367-1955 robert.varley@lmco.com

Charles Vinick Alliance to Protect Nantucket Sound 4 Barnstable Rd. Hyannis, MA 02601 508-775-9767 cvinick@saveoursound.org

Nancy Wallace NOAA, National Ocean Service 1305 East West Hwy. Silver Spring, MD 20910 301-713-3070 nancy.wallace@noaa.gov William Toman SAIC 901 D Street, SW, Suite 201 Washington DC 22024 703-625-9242 william.toman@saic.com

David G. Tuerck Beacon Hill Institute Suffolk University, 8 Ashburton Place Boston, MA 02108 617-573-8750 dtuerck@beaconhill.org

Sally Valdes MMS 381 Elden Street Herndon, VA 20170 703-787-1707 sall.valdes@mms.gov

Richard R. Veit City University of New York/College of Staten Island 2800 Victory Blvd. Staten Island, NY 10314 718-982-3853 veitrr2003@yahoo.com

Barbara Wallace MMS 381 Elden Street, MS 4041 Herndon, VA 20170 703-787-1512 barbara.wallace@mms.gov

William Waskes MMS 381 Elden Street Herndon, VA 20170-4871 703-787-1287 will.waskes@mms.gov Geoffrey Wikel MMS 381 Elden Street Herndon, VA 22071 703-787-1283 geoffrey.wikel@mms.gov

Chris Wissemann Winergy, LLC 150 Motor Parkway, Suite 425 Hauppauge, NY 11788 631-434-9100 chris@winergyllc.com Judy Wilson MMS 381 Elden Street, MS 4040 Herndon, VA 20170 703-787-1075 judy.wilson@mms.gov

Thomas C. Woodworth MMS 2121 Jamieson Ave., #1009 Alexandria, VA 22314 703-787-1198 thomas.woodworth@mms.gov

Appendix C

Workshop to Identify Alternative Environmental Information Needs

Presentation Slides for Invited Speakers





Biography

- Kurt E. Thomsen
- Crane operator/assessor
- Construction architect
- Bs in Business Management
- Founded A2SEA in 2000 based on Patent aplication filed in Sept 1999. Now world leading installation contractor
- Owner AOS founded in 2006

Clients represented Cape Wind Associates Dong Energy DOTI GmbH Con Energy Projects Sesent Wind Nai Kun N









Waterdepth and technology

- 2002: Waterdepth 15 m
- 2007: Waterdepth 27 m
- 2012: Waterdepth 39 m?
- Probably not because: 2002: Turbines are 2 MW

2007: Turbines are 3 MW 2012: Turbines are 3.6 MW Output of the turbines are not enough to cover the add cost and cable loss e foundation and installation technology will be to

Wave and wind climate

- The wind should blow on average 9 m/s
- The waves should not exceed 2 m Hs in more than 5% of the year
- · The currents should not be too large as seabed moves
- Thus a sheltered area should be sought out
- Shallow water shoals are desirable

Customers

- The windfarm should be close to a loadcentre because
- The power should run as short a distance as possible due to transfer losses
- Windpower should be generated where it is needed, ie. Close to densly populated areas
- This make windfarms work costeffectively

Nature

- When all objectives are met the windfarm
- should also be placed sensibly
- We know that:
- · Birds avoid windfarms offshore
- Fish are attracted to the foundations which create artificial reefs and nesting grounds
- Seals and mammals don't really care
- The impact of an offshore windfarm is minimal to wildlife in general

Current state in offshore wind

- · Turbines are still in the training years
- The size hasn't significantly grown in the first 7 years
- The reliability of proven turbines is there
- Foundation cost hasn't significantly dropped
- Cabling is still a very big challenge

Current state of technology

- Siting in shallow waters is crucial
- The second generation is not proven

waterdepths

ABIZ Contractor

- Installation contractors can not cope with larger turbines and foundations at the moment
- Foundation design is not geared for large

What we know

- Turbines and foundations can be installed succesfully
- Windfarms can operate in coexistence with wildlife
- Cost can be comparable to "normal" power
- The windfarms offshore work at a very high reliability in general



What they also do

- They generate clean power for your neighbourhood
- They offset CO2 emissions from day one
- They create opportunity for wildlife above and below sealevel
- As a consequence they generate opportunity for the users of the offshore environment



The future trends

- The turbines will become larger
- They will move further offshore
- They will generate more power
- They will become even more competitive towards "normal" power production

-

Larger turbines

- Development is mostly prohibited by:
- Current state of supporting technologies
- Shortage in supply

- Shortage in contractor capacity
- Shortage in contractor capability

Further offshore?

- This is mainly prohibited by:
- Distance vs. Powerloss
- Foundation cost vs. Output
- Downtime and repair of turbines vs average weather conditions
- Cost of supporting plant and personnel
- Overall safety of supply, repair and maintenance

When will this happen

• An example: Airbus 380 vs. Boeing 747

• Question: Why was the jumbo jet the biggest for over 40 years?

 The answer: Lack of supporting technologies

My forecast

- The 5 MW turbine will mature over the next 5 8 years and become an industry runner
- This will facilitate development of windfarms which become more efficient than they already are, but not necessarily further offshore
- The offshore contractors will see a market for these turbines within the next three to five years
- So by 2015 we will no longer see the 3 MW turbines offshore

The permitting process,

- The siting of an offshore windfarm should reflect what we want to achieve – clean efficient power
- The permitting process should adress the information which is needed, not what is nice to have. Ask for information which helps the process rather than delay it
- Be realistic in the scoping of EIA and other documents



What is the next step

- To finish the regulatory framework, taking advantage of the lessons learned in Europe
- To finish the permitting process for the first offshore windfarm(s) in the USA
- To start building for a cleaner future offshore.

























Aim to bring down financial risk

- Screening for site suitability (2003)
- Tender: Fixed price in 50.000 full load hours
- TSO to finance, construct and operate transformer
- TSO obliged to connect wind power and expand grid if
- Security that grid connection is available in due time

Ë N 🖥 R G Y

- Financial compensation if the power produced is
- One stop shop communication

A one stop shop – Streamlining consent procedures > The Danish State has all competence within the 12 NMZ and in the Danish EEZ 2 Delegation to the DEA of juridical authority from the minister of Transport and Energy > Pre-investigation and exploitation of energy at sea > Construction of electricity production plants and grid-connections at sea > Off-shore wind-power plants is thus consented and approved by the Danish Energy Authority in co-operation with other authorities - A one stop shop procedure Terms: > In pursuance of the electricity act > In pursuance of legislation from other authorities > On the cause of EIA suggestions and objections

ENERGY





3

Conclusions from the Environmental programme

- Public acceptance
- Migrating birds avoid collision
- Seals behaviour not affected
- Harbour porpoises return to the sites
- Significant artificial wreck effect observed







<section-header><section-header> <section-header> Conclusions Perception • Sibility - large wind turbines, • Conomic decline on local level - fishery and tourists • Environmental issues - birds • Divinonmental issues - birds • Divinonm





























Fish

Results:

- Same amount of sand eel in- and outside wind farm
- Artificial reef effect methodological challenge
- Electromagnetic field methodological challenge









































of the United States.'















































National Avian – Wind Power Planning Meeting I July 1994

Meeting Outcome: Five Major Research Areas

- Assess mortality attributable to wind turbines at existing sites (including control data from "no turbine" sites)
- Predict mortality at planned wind power sites, based in part on previous bullet
- Predict population consequences
- · Identify ways to reduce bird kills at wind plants
- · Set values for off-site mitigation

























Ocean Renewable Device Types

Ocean energy type	Technology types	Estimated global resource
Ocean wave	Attenuator, Collector, Overtopping, OW OWSC, Point absorber, Submerged pre differential, Terminator, Rotor	NC, 8 000-80 000 TWh/year ressure
Tidal current	Horizontal/Vertical-axis turbine, Oscill hydrofoil, Venturi	llating 800+ TWh/year
Salinity gradient	Semi-permeable osmotic membrane	2 000 TWh/year
OTEC	Thermo-dynamic ranking cycle	10 000 TWh/year
53	25 Cosan Wave Tdal Current Tdal Current OTEC Salinity Oradient	 81 wave, tidal, OTEC, and salinity devices in development worldwide 2x industry growth from 2003 to 2006 Only 14 full scale devices deployed at sea. Only 12 in the USA
Ocean Ene	ergy Project Types (%)	Only 3 in the USA NPEL National Toronalds Process Laboratory

Concluding Remark

World-wide electrical energy consumption is projected to grow by about 75% over the next 20 years. All energy technologies have some environmental impacts. Wind and ocean energy are developing rapidly, and a modest investment in environmental R&D now could make the impacts negligible. This would give us a carbon free electricity generating choice that could meet at least 20% of the world's energy needs.



NREL Avian Studies Available at: http://www.nrel.gov/wind/avian_lit.html

- Permitting of Wind Energy Facilities: A Handbook
- A Pilot Golden Eagle Population Study in the Altamont Pass Wind Resource Area, California
- A Population Study of Golden Eagles in the Altamont Pass Wind Resource Area, Second-Year Progress Report
- Ponnequin Wind Energy Project Reference Site Avian Study
 A Population Study of Golden Eagles in the Altamont Pass Wind Resource Area: Population Trend Analysis 1994-1997
- Predicting the Response of Bird Populations to Wind Energy-Related Deaths
- The Response of Red-Tailed Hawks and Golden Eagles to Topographical Features, Weather, and Abundance of a Dominant Prey Species at the Altamont Pass Wind Resource Area, California, April 1999-December 2000
- Searcher Bias and Scavenging Rates in Bird/Wind Energy Studies
- Status of Avian Research at the National Renewable Energy Laboratory (2001)
- Status of the US Dept. of Energy/NREL Avian Research Program (1999)
- Studying Wind Energy/Bird Interactions: A Guidance Document

ANREL NO.

9

Offshore Wind European Environmental References

- European Union, COD, Principal Findings 2003-2005, prepared by SenterNovem, Netherlands, www.offshorewindenergy.org
- Offshore Wind: Implementing a New Powerhouse for Europe, Greenpeace International, March 2005 http://www.greenpeace.org/international/press/reports/offsh
- ore-wind-implementing-a
 Danish (Horns Rev and Nysted) Ecological Studies
 http://www.hornsrev.dk/Engelk/default_ie.htm
 and
 http://wk.nystedhavmoellepark.dk/frames.asp?Page_ID=44&
 Page Ref=44&Templates ID=1
- U.K.'s Strategic Environmental Assessment
 <u>http://www.og.dti.gov.uk/offshore-wind-</u>
 sea/process/envreport.htm













Why Go Offshore?!

- Limited space on land
- Possibility of being close to centres of demand
- Better wind resource, higher quality wind
- Bigger projects possible with larger turbines
- Suitable seabed conditions are required
- Foundations are key
- Diversification of offshore industries away from more traditional areas

BWEA





The National Support Structure

- COWRIE
 Collaborative Offshore Wind Research Into the Environment
- RAG Research Advisory Group
- OREEF
 Offshore Renewable Energy and Environment Forum
- NOREL Nautical Offshore Renewable Energy Liaison Group
- FLOWW Endbrace
 Fisheries Liaison for Offshore Wind and Wet
 BWEA











































Licence Application Process our advisers

- Centre for Environment Fisheries and Aquaculture Science (Cefas) Aquatic scientific research and consultancy centre. Optimise resource development and utilisation Understand and assess environmental impacts Minimise environmental costs of marine activiti
- Joint Nature Conservation Committee (JNCC) JNCC delivers the UK and international responsibilities of the country nature conservation agencies Natural Englind. Advise Government on the development and implementation of preference of the fetching, nature conservation in the UK and preference of the fetching, nature conservation in the UK and
- internationally: Provide advice and disseminate knowledge on nature conservation issues affecting the UK and internationally. Establish common standards throughout the UK for nature conservation, including monitoring, research, and the analysis of results;
- Commission or support research.
- Natural England

 Conservation of England's natural environment.

 Enjoyment and understanding of the natural environment.

 Sustainable use and management of the natural environment.

 Decisions that collectively secure the future of the natural
 nt



Protecting the environment monitoring conditions

- Marine Pollution Contingency plans for spills
 and collision incidents
- Liaison officers to maintain communication between licence holder, contractors, fishermen and conservation groups

- Marine Marmal Observers & protocol to mitigate potential impacts
 Attenuation of electro-magnetic field strengths associated with cables
 Monitoring programme (birds, benthic organisms, fish, noise and vibration) agreed before construction commences
- · Licence conditions: http://www.mceu.gov.uk/MCEU_LOCAL/FEP A/NEWSITEMS/LondonArray-letter.pdf

Case Study: Making sure licence conditions are enforceable LESSON LEARNT: MAKE SURE LICENCE CONDITIONS CLEARLY SPECIFY REQUIREMENT AND ARE TIME LIMITED

Is suc Interpretation of licence condition to implement measures to ensure safe navination

- Situation Number Wind fam developer and harbour authority have conflicting views: - Hator Arbority: Instituting strikes, will desure immers rate experiment immediately - Wind fam developer: confliction net fine limited. Temporary subulon agreed subject to there developer: confliction net mile limited. Temporary subulon agreed subject to there developer: confliction net mile limited. Temporary subject to there developer: confliction net mile limited. Temporary subject to there developer: confliction net mile limited. Temporary subject to there developer: confliction net mile limited. Temporary permanent solution found

Result Meeting called to discuss situation Harbour Authority tasked with implementing ter Both to sign up to permanent solution














































































Lost in translation	RPS Energy
100MW Huittengxile Wind Farm, Inner Mongolia	
EIA, January 2005 (28 pages, funded by the World Bank)	
wind turbines are so obvious that the birds can see it clearly and avoid it	
it will form beautiful and attractive scenery. In the vast grassland, a lot of white wind turbines stand in order and rotate in the blue sky, the white cloud and green grassland	
the number of wind turbines will grow and the scenery will be surely more beautiful	



Wave Energy Development in the U.S. – Present Status and Future Trends

Minerals Management Service

Workshop to Identify Alternative Energy Environmental Information Needs Herndon, VA

26 June 2007

George Hagerman

Virginia Coastal Energy Research Consortium Virginia Tech Advanced Research Institute

Presentation Outline

Resource characteristics

Governed by local winds and offshore storms

U.S. production potential

- 250-260 TWh per year (EPRI, 2004)
- Comparable to annual energy output of all existing conventional hydro-electric projects in US

General types of conversion technology

 Highly diverse alternatives; classified into Terminators, Attenuators, and Point Absorbers

Conversion technology status

Has yet to converge on single best technical approach (if such exists)



Winds Move ~60% of Excess Solar Energy from Equator to Poles



Global, High-Altitude Wind Energy Flux Distribution

Wind energy flux at 550 hPa (-1.500 m elevation); scale below in wats persa m 500 1000 1500 2000 2500 3500 4500 4500 5000 The annual average wind power density at -1.6 km above sea level (elevation of the so-called 'gradient wind') is 110 5 kilowatts per square meter of cross-sectional area

Wind Over Water Generates Waves





Wave Energy Flux in Typical U.S. Mid-Atlantic Sea State



Global Wave Energy Flux Distribution

U.S. Offshore Wave Energy Resources









OWC Terminator: Onshore LIMPET



<section-header><complex-block>

Overtopping Terminator: Wave Dragon



Wave Dragon Prototype Trials

Prototype is 58 m wide (between tips of funneling side walls) and 33 m long, with a reservoir volume of 55 m³ and a displacement of 237 metric tons. Total rated capacity is 17.5 kWe.



Funneling side walls are moored separately from central floating reservoir.



Pelamis Engineering Development



1998 – 2003: 1/20 and 1/33-scale models tested to physically validate numerical simulations of wave energy absorption efficiency and mooring loads (survivability)







2002 – ongoing: Full-scale power module bench rig tested to qualify mechanical and electrical components and to assess MTBF (reliability) and control system performance

Pelamis Sea Trials and Pilot Plant





AquaBuOY1 MW Project to be Installed off Makah Bay, Washington



Point Absorber: OPT PowerBuoy

- OPT's PowerBuoy™ system extracts the natural energy in ocean waves, and is based on the integration of patented technologies in hydrodynamics, electronics, energy conversion and computer control systems. The PowerBuoy is a *smart' system capable of responding to differing wave conditions.
- The rising and falling of the waves off shore causes the buoy to move freely up and down. The resultant mechanical stroking is converted via a sophisticated power take-off to drive an electrical generator. The generated power is transmitted ashore via an underwater power cable.

OPT website June 2007





www.oceanpowertechnologies.com

Navy Funded Demonstration Project at Kaneohe Marine Base, Oahu, Hawaii







US Wave Energy Projects in FERC Regulatory Process

- Finavera AquaBuOY
 Makah Bay, WA demonstration project entered FERC alternative licensing process in September 2003; fabrication underway now, summer 2008 deployment
 - Coos County, OR preliminary permit issued
 - · Humboldt County, CA preliminary permit applied for

OPT PowerBuoy (including OPT wholly-owned subsidiaries Oregon Wave Energy Partners)

- Two Oregon projects with preliminary permits issued (Reedsport and Coos Bay)
- Two Oregon projects with preliminary permits applied for (Newport and Fairhaven)

- Type of device yet to be determined Douglas County, Oregon (preliminary permit issued) Lincoln County, Oregon (preliminary permit applied for)
 - Two PG&E "WaveConnect" Projects in Northern California



























Major Issues – by Stressor • Wave and current modification • New hard structures – water column and benthic • Electromagnetic fields • Chemical toxicity • Acoustics • Cumulative effects





























<section-header><section-header><section-header><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text>





Summary – Take Home

- Many generic effects are known and may be roughly predicted or modeled
- Effects will be time- and location-specific
- Some effects will be technology-specific
 Vertebrate behavior/responses are complex
- and may be species-specific
- Major information gaps need to be filled
 Some forms of mitigation are expected to be
- effective
 Siting is particularly important for array effects
 Cumulative effects will be quite difficult to
- predict
- User conflicts may be more likely than ecological effects to limit development

New Responsibilities: Energy Policy Act of 2005



Gave DOI comprehensive authority to manage future development of promising new ocean energy sources in the OCS (wind, wave, ocean current, and solar) and responsible alternate use of OCS facilities



Build New Regulatory Framework

Using:

- Meaningful dialogue and partnerships with stakeholders
- Marine regulatory experience
- Sound science, engineering and environmental protection principles



Program & Regulatory Development

Two provisions under Section 388:

- Production, transportation, or transmission of energy from sources other than oil and gas [Alternative Energy]
- Use of currently or previously OCSLA-authorized facilities for energy-related purposes or for other authorized marine-related purposes [Alternate Use]

Dialogue and Partnerships with Stakeholders

Held stakeholder meetings

- Identify stakeholders
- Describe key issues and concerns
- Characterize energy needs and trends
- Describe current and future technology development
- Identify State/Local regulations

Kay Regulatory Considerations

- Safety
- Protection of the environment
- Coordination with affected State & local governments and Federal agencies
- Fair return for use of OCS lands
- Equitable sharing of revenue with States

Competition Requirement

Secretary shall issue a lease, easement, or right-of-way on a competitive basis unless the Secretary determines after public notice of a proposed lease, easement, or right-of-way that there is no competitive interest.

Major Regulatory Elements*

- Coordination
- Lease Issuance Competitive & Noncompetitive Coordination
- Lease Administration **Bonding & Payments**
- Project Plan Reviews Site Assessment & Construction and Operations
- Conduct of Approved Plan Activities Installation, Production Environmental and Safety Monitoring & Inspections
- Decommissioning *Includes compliance with Federal statutory requirements (e.g., NEPA, CZMA, ESA, MMPA, CWA, CAA etc.)

Programmatic EIS

- Involved the public early in identifying issues of concern and interest
- Supports implementation of a new Federal program
- Identifies generic impacts of alternative energy and alternate use of existing facilities
- Recommends mitigation measures

Scope of the Programmatic EIS

- Timeframe Over the next 5-7 years
- Technologies Wind, Wave, Ocean Current
- Geographic Location East, West, and Gulf Coasts
- Federal Waters greater than 3 nm, but water depths up to 100 meters

Programmatic EIS & Rulemaking Targets

Spring 2007

- ✓ Published draft Programmatic EIS www.ocsenergy.anl.gov Held public hearings
- Late Summer 2007
 - Issue final Programmatic EIS
 - Publish Notice of Proposed Rulemaking and open public comment period



Sound Science

MMS uses environmental studies to inform our Program decisions

Develop Strategic Studies Plan

 Contracted with Research Planning, Inc. (RPI) to conduct a 9-month study entitled "The Worldwide Synthesis and Analysis of Existing Information Regarding Environmental Effects on Alternative Energy Uses of the Outer Continental Shelf"



- Identify data gaps and study needs
- Develop appropriate methods
- Suggest priorities for future
- studies

Sound Science

Final Strategic Studies Plan

- Use June Workshop results to prepare draft Alternative Energy Strategic Studies Plan
- Seek MMS Scientific Advisory
 Committee input and comments
- Serve as a guide for Alternative Energy Program environmental studies
- Use studies results to inform our leasing and plan review decisions

Manage Existing OCS Alternative Energy Projects

MMS will not issue decisions until the Alternative Energy Program is in place

Cape Wind Energy Project

Cape Wind Associates, LLC, wind park proposal in Nantucket Sound, MA, consisting of 130, 3.6 MW wind turbine generators, about 4.7 miles offshore

Draft EIS will consider a range of alternatives:

- no action
- phased build-out
- smaller configuration
- sites offshore RI, MA, & ME for comparisons Targets:
- Summer 2007 File draft EIS; open comment period
 - Fall 2007 Hold Public hearings

LIOWP Project

Long Island Power Authority and Florida Power and Light Energy propose to build offshore wind park about 4 miles off the south shore of Long Island, New York

Draft EIS will consider a range of alternatives:

- no action
- one alternative site off of Long Island
- one alternative site in deepwater
- one alternative site onshore
- Targets
 - TBD

Questions?

Maureen A. Bornholdt (703) 787-1300 www.mms.gov/offshore/ RenewableEnergy/ RenewableEnergyMain.htm



















 Integrated Licensing Process (ILP) Traditional Licensing Process (TLP) Alternative Licensing Process (ALP) Prefiling: Consult with 	Licensing	y Processes
Prefiling: Consult with Interested parties Determine study plan and conduct studies Postfiling: Solicit comments Do environmental analysis Make recommendation	Integrated Licensing — Traditional Licensing — Alternative Licensing	g Process (ILP) g Process (TLP) g Process (ALP)
Prepare license application to Commission Commission	Prefiling: • Consult with interested parties • Determine study plan and conduct studies • Prepare license application	Postfiling: • Solicit comments • Do environmental analysis • Make recommendation to Commission • Commission decision







- Finavera Renewables Ocean Energy, Ltd.
- Four, 250-kW wave buoys ("AquaBuOYs"), submarine transmission cable, and shore station
- AquaBuOYs located over state aquatic lands
- Facilities would be placed within boundaries of Olympic Coast Marine National Marine Sanctuary (NOAA) and Makah Indian Reservation, and near Flattery Rocks Wildlife Refuge (U.S. Fish and Wildlife Service)
- Annual generation estimate: 1,500 MWh



Makah Bay Wave Energy Project

- October 3, 2002: FERC determined license needed to construct and operate the proposed wave energy project
- September 4, 2003: FERC approved Finavera's request to use ALP to prepare license application
- September 2003: Initial scoping of issues, public meetings, and site visits
- October 2002-June 2005: Studies identified by stakeholder group and conducted by Finavera

Makah Bay Wave Energy Project

- Finavera conducted a number of studies, including the following:
- Current Analysis: magnitude, direction
- Surface Wind and Wave Analysis
- Sediment Analysis: grain size, type, etc.
- Seafloor and Vegetative Mapping
- Literature Review: Marine plant and animal species present in the project area

Makah Bay Wave Energy Project

FERC Environmental Assessment (May 2007) determined potential minor adverse effects on:

- Seabed, benthic organisms, marine fish and mammals, and turbidity during construction
 Fish, marine mammals, and seabirds during
- Fish, marine mammals, and seabirds during operations (e.g., EMF, noise, entanglement, derelict fishing gear and other debris, etc.)
- Long-term loss of small amount of fishing and crabbing grounds within project exclusion zone

Makah Bay Wave Energy Project

Areas where specific information for the EA was less abundant:

- Noise assessments specific to wave energy technologies
 - EMF assessments specific to wave energy generation, including unburied submarine transmission lines
- Marine mammal and seabird entanglement and collision with buoys and mooring lines



- Cultural resources monitoring during
- construction
 Recreational use monitoring (buoy array as a tourist attraction)

Contacts

- http://www.ferc.gov/industries/hydropower.asp (Look for link to "hydrokinetics.")
- Stephen Bowler <u>stephen.bowler@ferc.gov</u>
- Nick Jayjack <u>nicholas.jayjack@ferc.gov</u>

Appendix D

Workshop to Identify Alternative Environmental Information Needs

Biographies for Invited Speakers

Kurt Thomsen Managing Director, Advanced Offshore Solutions, ApS

Kurt E. Thomsen is a 44 year old construction architect who in 1999 filed a patent application for the first dedicated offshore wind farm installation vessel. Following the patent application he formed the company A2SEA which today is one of the leading installation companies in the offshore wind industry. Kurt E. Thomsen and A2SEA have installed more than 75% of all offshore wind farms in the world and today are the leading repair as well as operations and maintenance companies.

Today A2SEA is an internationally recognized leader in the industry with more than 130 employees and three turbine installation vessels in their fleet. Following the successful development and growth of the company, Kurt E. Thomsen in 2006 decided to leave A2SEA and start up a consultancy company for the offshore wind industry.

Working from his office in Århus, Denmark, he has advised clients such as Cape Wind Associates, DONG Energy, Eon, EWE, Essent Renergys and Vattenfall—companies which are devoted to installing large scale offshore wind farms throughout Europe and the USA. Currently Kurt E. Thomsen is working on five offshore projects and advising on transport and installation logistics for the abovementioned companies.

Steffen Nielsen, Ph.D. Head of Section, Danish Energy Authority

Master in Environmental Planning 1993 - thesis in technology transfer of wind-power to India – Roskilde University, Denmark.

Ph.D. in Social Science 1999 - thesis in climate change mitigation and environmental sustainable solutions in developing countries, Risoe National Laboratory, UNEP, Rescaled University. With the Danish Energy Authority since 1998 doing:

- Policy design for renewables in the liberalised electricity marked, e.g. feed-in tariff assessments and preparation of the amendments for the electricity law in Denmark.
- Consent and approval of all offshore wind power in Denmark since the Middelgrunden project 1999.
- Coordinated the Danish environmental monitoring programme for large scale offshore wind-power at Horns Rev and Nysted.
- Project management of a European Policy Seminar on Offshore Wind Power, producing the "Copenhagen Strategy 2005."
- Chairman for the committee for future offshore wind power development in Denmark, reported 2007.
- Speaker and chair at conferences, seminars, and workshops in Denmark and abroad.

Since 2002, officially appointed examiner at the Department of Environmental, Social and Spatial Change, Roskilde University.

Bob Thresher, Ph.D. National Renewable Energy Laboratory

Dr. Thresher is the Director of the National Wind Technology Center in Golden, Colorado, a division of the U.S. Department of Energy's National Renewable Energy Laboratory (NREL). He earned a tenured professorship in Mechanical Engineering at Oregon State University where he taught courses in Applied Mechanics and initiated pioneering researcher in the mechanics of Wind Energy Systems during the 1970s and early 80s. He joined NREL in 1984 and has provided leadership for the growth of NREL's wind program from \$5MM/year at its inception, to its current level of about \$30MM/year. He has published extensively and is recognized internationally as one of the leading experts in research, development, and commercialization of wind technologies. He also serves as a member of the Advisory Panel on Ocean Energy Technologies for the Electric Power Research Institute's. In 2005, Dr. Thresher testified before the U.S. Senate Committee on Energy and Natural Resources on the research and development needed to accelerate the use of Wind and Ocean Technologies on the U.S. OCS.

Michael Hay Head of Offshore Renewables, British Wind Energy Association

Mr. Hay is Head of Offshore Renewables at the BWEA. He is a graduate of Imperial College London with an MSc in Environmental Technology and Energy Policy where his research project was concerned with the potential role for UK regions in the development of a marine renewables industry. On completion of this work he joined the Renewables Innovation Review Team at the Department for Trade and Industry. This small external group assessed all renewable energy technologies, including wave and tidal stream, and advised Treasury on the allocation of funds in the Governments 2004 Comprehensive Spending Review. In February 2004 Mike joined BWEA in a new position that expanded its hugely successful wind mandate in order to champion the emerging wave and tidal stream energy sector. As well as co-ordinating the interests of BWEA's 117 marine members, Mike also leads the Association's offshore wind work, representing these industries in discussions with Government, offshore stakeholders, and the media.

Rachael Mills Dept. for Environment, Food and Rural Affairs (DEFRA), United Kingdom

Rachael Mills joined the UK's Department for Environment, Food and Rural Affairs (Defra) in November 1997, spending her first few years in the Department's science group where she led work to develop knowledge transfer practices in the agricultural sector. Her Defra career was largely project management based, and she led some major projects to underpin Government policy. In 2004 she was seconded to the Department for International Development to establish best practice project management techniques within the department and to manage a Programme Office to oversee several cross departmental projects. Rachael took up the post to lead offshore renewable energy licensing within Defra in October 2006. The small team is responsible for implementing an efficient license delivery and management service for offshore renewables. In April this year, the team moved to the Marine and Fisheries Agency (an executive agency of Defra). The Marine and Fisheries Agency operates a network of enforcement, scientific, and administrative staff who carry out a range of statutory duties including enforcing sea fisheries legislation. Rachael has an Honors Degree in Zoology from the University of Liverpool (UK).

Chris Jenner Technical Director, Renewables, RPS Group Plc

Mr. Jenner is Technical Director with RPS with specific responsibility for the marine renewables business. Chris has over 10 years experience in environmental assessment and management in UK and overseas, more recently he has spent the last 6 years working for the offshore wind industry for a number of Round 1 and Round 2 projects and proposals in U.S. and Canada. Areas of interest include environmental assessment, offshore surveys, geological seabed risk assessment, and consents compliance during construction. Chris was responsible for preparing the Environmental Statements and coordinating surveys for the Gunfleet Sands 1 & 2, London Array, and Lincs projects, peer reviewed the Sheringham Shoals and Walney Offshore Wind submissions, and is currently engaged by E.ON and Lunar Energy to manage the EIA process for future tidal stream energy projects. His current projects include acting as Consents & Environment Manager on behalf of Centrica Renewable Energy Ltd. for the Lynn and Inner Dowsing – the first offshore wind project currently under construction in The Wash.

George Hagerman Virginia Tech Advanced Research Institute

Mr. Hagerman has over 25 years experience evaluating and optimizing the design, performance, and economics of renewable ocean energy conversion systems, including offshore wind energy, wave power, tidal power, and ocean thermal energy conversion (OTEC). He is a research faculty member at the Virginia Tech Advanced Research Institute in Arlington, Virginia, and Research Director for the newly established Virginia Coastal Energy Research Consortium, a multi-university partnership exploring offshore wind power, hybrid wind-wave power systems, and fuels derived from marine biomass as potential future energy supply alternatives for Virginia. Under the Electric Power Research Institute's collaborative wave energy and tidal stream energy feasibility studies, he was responsible for resource assessment, site characterization, and environmental issue identification in six U.S. states and two Canadian provinces.

Mr. Hagerman has a Master of Science in Marine Sciences and a Bachelor of Science in Zoology, both from the University of North Carolina at Chapel Hill. He has published more than twenty professional papers on ocean energy resources and technologies, has edited ocean energy articles and reports for a variety of non-governmental organizations, and has testified before several Congressional committees on pending ocean energy legislation.

Greg McMurray, Ph.D. Ocean and Coastal Management Program, Oregon Department of Land Conservation and Development

Dr. McMurray was born and raised in the vicinity of Cleveland, Ohio, where he loved to catch frogs. He attended Ohio University, obtaining his Bachelor's degree in zoology in 1969, and spent his first year thereafter working in fish production for the Ohio Division of Wildlife. With a renewed interest in aquatic ecology, he earned a master's degree in biology (limnology) at the University of Akron, and came to Oregon where he completed his Ph.D. in biological oceanography at Oregon State University in 1977. After a year of postdoctoral research in San Francisco Bay with the U.S. Geological Survey, he spent six years in environmental consulting, involved largely with marine/estuarine environmental baseline assessments in the Misty Fjords National Wilderness Area, southeast Alaska. From 1984 until 1991, he continued work on marine environmental assessments for the Oregon Department of Geology and Mineral Industries, including exploration and biological baseline programs for ocean spreading center hydrothermal sulfide minerals on Gorda Ridge and nearshore placer minerals offshore southern Oregon. Dr. McMurray also spent a brief time as a technical director for Exxon USA's natural resource damage assessment studies in Valdez, Alaska, during 1989. From 1991 until 2004, he was a Principal Environmental Analyst with the Oregon Department of Environmental Quality, working much of that time as the Program Coordinator of the Pacific Northwest Coastal Ecosystems Regional Study (PNCERS), which investigated the relationships of natural versus human forcing in Oregon and Washington estuaries. He is presently the Marine Affairs Coordinator for the Oregon Department of Land Conservation and Development, where his principal duties are to staff the state's Ocean Policy Advisory Council and to facilitate the availability and use of science to inform marine natural resource management issues. Dr. McMurray and his wife, Mary, live on the edge of a swamp in Lake Oswego with their daughter, Lela, and son, Russell, where they all love to listen to the northern chorus frogs.